Narendra Tuteja Sarvajeet Singh Gill *Editors*

Crop Improvement Under Adverse Conditions



Crop Improvement Under Adverse Conditions

Narendra Tuteja • Sarvajeet Singh Gill Editors

Crop Improvement Under Adverse Conditions



Editors Narendra Tuteja, Ph.D. Plant Molecular Biology Group International Centre for Genetic Engineering & Biotechnology Aruna Asaf Ali Marg New Delhi 110067 India narendra@icgeb.res.in

Sarvajeet Singh Gill, Ph.D. Stress Physiology & Molecular Biology Laboratory Centre for Biotechnology Maharshi Dayanand University Rohtak 124001, Haryana India ssgill14@vahoo.co.in

ISBN 978-1-4614-4632-3 ISBN 978-1-4614-4633-0 (eBook) DOI 10.1007/978-1-4614-4633-0 Springer New York Heidelberg Dordrecht London

Library of Congress Control Number: 2012954541

© Springer Science+Business Media New York 2013

This work is subject to copyright. All rights are reserved by the Publisher, whether the whole or part of the material is concerned, specifically the rights of translation, reprinting, reuse of illustrations, recitation, broadcasting, reproduction on microfilms or in any other physical way, and transmission or information storage and retrieval, electronic adaptation, computer software, or by similar or dissimilar methodology now known or hereafter developed. Exempted from this legal reservation are brief excerpts in connection with reviews or scholarly analysis or material supplied specifically for the purpose of being entered and executed on a computer system, for exclusive use by the purchaser of the work. Duplication of this publication or parts thereof is permitted only under the provisions of the Copyright Law of the Publisher's location, in its current version, and permission for use must always be obtained from Springer. Permissions for use may be obtained through RightsLink at the Copyright Clearance Center. Violations are liable to prosecution under the respective Copyright Law.

The use of general descriptive names, registered names, trademarks, service marks, etc. in this publication does not imply, even in the absence of a specific statement, that such names are exempt from the relevant protective laws and regulations and therefore free for general use.

While the advice and information in this book are believed to be true and accurate at the date of publication, neither the authors nor the editors nor the publisher can accept any legal responsibility for any errors or omissions that may be made. The publisher makes no warranty, express or implied, with respect to the material contained herein.

Printed on acid-free paper

Springer is part of Springer Science+Business Media (www.springer.com)



Professor Arturo Falaschi (January 21, 1933–June 1, 2010)

Prof. Arturo Falaschi was born in Rome and graduated in Medicine in 1957 from Milan University. He undertook post doctoral trainings with J. Adler and Har Gobind Khorana (Nobel Prize in 1968 for deciphering the genetic code) in Wisconsin, USA (1961–1962), and later with Arthur Kornberg (Nobel Prize in 1959 for his studies on DNA replication) at Stanford (1962–1965). His main field of research was in the field of DNA replication and his contributions in this field are significantly important. His scientific work featured in the most prestigious international journals. Prof. Falaschi remains one of the few international researchers whose scientific activity is

documented throughout almost fifty years (from 1962 to 2010). Prof. Falaschi was responsible for the establishment of several research institutes and was a strong believer in the internationalization of science. Prof. Falaschi was verv articulate and convinced several governments in the developed and the developing world to establish a 3-component International Centre for Genetic Engineering and Biotechnology (ICGEB), with one component in New Delhi, India. one in Trieste, Italy and one in Cape Town, South Africa. All these centres are devoted to research and training of young researchers from the developing world (www.icgeb. org). One of the focuses of ICGEB, New Delhi component is the development of crops resistant to various stresses. Prof. Falaschi was the mind and driving force in the founding and development of ICGEB, where he served as the Director General from 1989 to 2004. From 2004 to 2010. he worked as ICGEB Distinguished Scientist and Professor of Molecular Biology, Scuola Normale Superiore SNS, Pisa, Italy.

This book is dedicated to the memory of Prof. Arturo Falaschi as a token of our appreciation and respect for him and his achievements.

Foreword

Plants are fundamental to life on Earth and they have been harnessed by humans for food, feed, fibre, fuel and fun. The need to increase crop production is becoming more urgent due to increasing population and diversion of crops to biofuels production. Furthermore, this needs to be done sustainably with reduced inputs and in the face of global environmental change. It is also notable that one-third of the world's food production is estimated to grow under irrigation—much of this irrigation is unsustainable, using water supplies that are overexploited and under threat from changing weather patterns resulting from global climate change. It is estimated that to meet the recent Declaration of the World Summit on Food Security target of 70 % more food by 2050, an average annual increase in production of 44 million metric tons per year is required, which is a 38 % increase over historical increases in production.

The gap between potential yield and actual yield is primarily due to the effects of abiotic stresses on crop production. It is therefore an imperative to improve our ability to maintain crop production in environments with suboptimal conditions, such as low water or nutrient supplies, or high salinity. This is, of course, required in addition to improving the efficiency of delivery of existing technologies into developing countries through improved education and outreach.

As such, the book edited by Dr. Narendra Tuteja and Dr. Sarvajeet Singh Gill provides a useful and timely compilation of up-to-date overviews of advances in the important area of plant sciences, "Crop Improvement Under Adverse Conditions". In this volume, a range of papers have been brought together which address both the technologies required to understand mechanisms of abiotic stress tolerance and the biological questions to which those technologies need to be applied. An understanding of the molecular and physiological aspects of plant function is provided in this book, and the emphasis on contributors from developing countries is very valuable—delivery of improved technologies and improved varieties of crops in such regions will have the greatest relative impact on global food production.

The editors and contributors are to be congratulated on their efforts, and readers are recommended to use this volume for a long time to come.

Mark Tester Adelaide, Australia

Preface

Plant development and productivity are negatively regulated by various environmental stresses. Abiotic stress factors such as heat, cold, drought, salinity, wounding, heavy metals toxicity, excess light, flooding, high speed wind, nutrient loss, anaerobic conditions and radiations etc. represent key elements limiting agricultural productivity worldwide. The loss of productivity is triggered by a series of morphological, physiological, biochemical and molecular stress-induced changes. Such an unfavourable situation is in contrast with the increasing global food demand. World population is increasing at an alarming rate and is expected to reach more than nine billion by the end of 2050, whereas, plant productivity is being seriously limited by various abiotic stresses all over the world. Global climatic pattern is becoming more unpredictable with increased occurrence of drought, flood, storm, heat waves, and sea water intrusion. It has been estimated that abiotic stresses are the principal cause for decreasing the average yield of major crops by more than 50 %, which causes losses worth hundreds of millions of dollars each year. Therefore, to feed the world population maintaining crop productivity even under unfavourable environment is a major area of concern for all nations. Developing crop plants with ability to tolerate abiotic stresses is need of the day which demands modern novel strategies for thorough understanding of plant's response to abiotic stresses. Molecular breeding and genetic engineering have significantly contributed to expand the basic knowledge of the cellular mechanisms involved in stress response, suggesting new strategies to enhance stress tolerance.

In this book "Crop Improvement Under Adverse Conditions", we present a collection of 17 chapters written by 55 experts in the field of plant abiotic stress tolerance and crop improvement. It is a timely contribution to a topic that is of eminent importance. The chapters provide a state-of-the-art account of the information available on abiotic stress tolerance and crop improvement. In this book, we present the approaches for crop improvement under adverse environmental conditions. Chapter 1 deals with the research, development, commercialization, and adoption of drought- and stress-tolerant crops, where the factors affecting adoption of stresstolerant crops by farmers are explored which includes complementary technologies, competing technologies, appeal to first-time users, distribution and timing of benefits to users, and social perceptions of the technology. Chapter 2 uncovers the impact of extreme events on salt-tolerant forest species of Andaman and Nicobar Islands. Chapter 3 deals with greenhouse gases emission from rice paddy ecosystem and their management. The plant development path of mitigating greenhouse gases (GHG) from agriculture cropping systems is not yet well established. Therefore aggressive research strategies and field validations are needed for establishing 'plant development' as a sustainable tool for GHG mitigation in agriculture sector. Chapter 4 covers remote sensing applications to infer yield of tea in a part of Sri Lanka. Chapter 5 deals with the polyamines contribution to the improvement of crop plants tolerance to abiotic stress, where, mechanism of action of polyamines to protect crop plants from challenging environmental conditions has been discussed. Chapter 6 discusses the overlapping horizons of salicylic acid in different stresses. In this chapter, the indigenous accumulation and overlapping roles of SA under different environmental and physiological conditions highlighting its recently updated roles and regulations in plants is discussed. Chapter 7 focuses on the effects of oxidative stress within the nuclear compartment where DNA becomes the main target of the highly toxic reactive oxygen species (ROS). Chapter 8 deals with a fast and reliable approach for crop improvement through in vitro haploid production. This chapter will act as a guide to prospective scientists working in the area of haploid production intended for crop improvement. Chapter 9 discusses the strategy for the production of abiotic stress-tolerant fertile transgenic plants using androgenesis and genetic transformation methods in cereal crops. Chapters 10 and 11 deal with the control and remedy of plant diseases through nanotechnology and the scope and potential of nanobiotechnology in crop improvement. The use of mutifunctionalised nanoparticles as plant transgenic vehicle for developing disease and stress resistant transgenic plants is discussed. Nanotechnological approaches on plants allow more efficient and sustainable food production by reducing the chances of disease and pest incidence in plants. In Chap. 11, thorough studies and reliable information regarding the effects of nanomaterials on plant physiology and crop improvement at the organism level are discussed. Chapter 12 deals with the role of nematode trapping fungi for crop improvement under adverse conditions. Chapter 13 uncovers the role of sugars as antioxidants in plants. This chapter discussed that soluble vacuolar carbohydrates (e.g. fructans) may participate in vacuolar antioxidant processes, intimately linked to the well-known cytosolic antioxidant processes under stress. All these insights might contribute to the development of superior, stress-tolerant crops. Chapter 14 deals with chromium toxicity and tolerance in crop plants, where, the mechanism of phytotoxicity and phytotolerance under Cr stress is discussed. Chapter 15 deals with boron toxicity and tolerance in crop plants, where, attempts to improve crop yields under boron-toxic soils is discussed. Chapter 16 deals with the approaches for stress resistance and arsenic toxicity in crop plants. Chapter 17 uncovers the mechanism of cadmium toxicity and tolerance in crop plants.

The editors and contributing authors hope that this book will include a practical update on our knowledge of "Crop Improvement Under Adverse Conditions" and lead to new discussions and efforts to the use of various tools for the improvement of crop plants under changing environment.

Preface

We are highly thankful to Dr. Ritu Gill, Centre for Biotechnology, MD University, Rohtak for her valuable help in formatting and incorporating editorial changes in the manuscripts. We would like to thank Prof. Mark Tester for writing the foreword and Springer Science+Business Media, LLC, New York, particularly Editor, Plant Sciences, Amna Ahmad and Developmental Editor/Project Manager, Daniel L.A. Dominguez and Andy Kwan, for their support and efforts. We have dedicated this book to Prof. Arturo Falaschi, the mind and driving force in the founding and development of ICGEB.

> Narendra Tuteja Sarvajeet Singh Gill

Contents

1	The Research, Development, Commercialization, and Adoption of Drought and Stress-Tolerant Crops Gregory Graff, Gal Hochman and David Zilberman	1
2	Impact of Extreme Events on Salt-Tolerant Forest Species of Andaman and Nicobar Islands (India) Alok Saxena, P. Ragavan and Mani Saxena	35
3	Greenhouse Gases Emission from Rice Paddy Ecosystem and their Management T. B. Dakua, L. Rangan and Sudip Mitra	65
4	Remote Sensing Applications to Infer Yield of Tea in a Part of Sri Lanka Saumitra Mukherjee, Jayasekara Balasuriya, Don Aruna, Pradeep Kumara and Chander Kumar Singh	91
5	Polyamines Contribution to the Improvement of Crop Plants Tolerance to Abiotic Stress Ana Bernardina Menéndez, Andrés Alberto Rodriguez, Santiago Javier Maiale, Kessler Margarita Rodriguez, Bremont Juan Francisco Jimenez and Oscar Adolfo Ruiz	113
6	Overlapping Horizons of Salicylic Acid under Different Stresses Mohd Irfan, Shamsul Hayat, Arif Shafi Wani and Aqil Ahmad	137
7	Genotoxic Stress, DNA Repair, and Crop Productivity Alma Balestrazzi, Massimo Confalonieri, Anca Macovei, Mattia Donà and Daniela Carbonera	153

Co	nte	nts
-0	me	iito

8	In Vitro Haploid Production—A Fast and Reliable Approach for Crop Improvement	171
	Rashmi Rekha Hazarika, Vijay Kumar Mishra and Rakhi Chaturvedi	
9	Production of Abiotic Stress Tolerant Fertile Transgenic Plants using Androgenesis and Genetic Transformation Methods in Cereal Crops S. M. Shahinul Islam and Narendra Tuteja	213
10	Plant Diseases—Control and Remedy Through Nanotechnology Remya Nair and D. Sakthi Kumar	231
11	Nanobiotechnology: Scope and Potential for Crop Improvement Faheem Ahmed, Nishat Arshi, Shalendra Kumar, Sarvajeet Singh Gill, Ritu Gill, Narendra Tuteja and Bon Heun Koo	245
12	Role of Nematode-Trapping Fungi for Crop Improvement under Adverse Conditions Rakesh Kumar Singh, Dipesh Kumar Trivedi and Amit Srivastava	271
13	Sugars as Antioxidants in Plants Darin Peshev and Wim Van den Ende	285
14	Chromium Toxicity and Tolerance in Crop Plants Ishrat Khan, Hema Diwan and Altaf Ahmad	309
15	Boron Toxicity and Tolerance in Crop Plants Robert J. Reid	333
16	Arsenic Toxicity in Crop Plants: Approaches for Stress Resistance Dhammaprakash Pandahri Wankhede, Meetu Gupta and Alok Krishna Sinha	347
17	Mechanism of Cadmium Toxicity and Tolerance in Crop Plants Sarvajeet Singh Gill, Naser Aziz Anjum, Ritu Gill, Mirza Hasanuzzaman, Pankaj Sharma and Narendra Tuteja	361
Inc	lex	387

xiv

The Editors



Narendra Tuteja was born in 1955. Currently, Dr. Tuteja is working as a Senior Scientist in Plant Molecular Biology Group, International Centre for Genetic Engineering and Biotechnology (ICGEB), New Delhi. Dr. Tuteja obtained his M.Sc., Ph.D and D.Sc. in Biochemistry from the Lucknow University in 1977, 1982 and 2008, respectively. He is fellow of the Academies of Sciences: FNASc. (2003), FNA (2007), FASc. (2009), FNESA (2009) and FTWAS (2012).

Dr. Tuteja has made major contributions in the field of plant DNA replication and abiotic stress signal transduction, especially in isolating novel DNA/RNA helicases and several components of calcium and G-proteins signaling pathways. Initially he made pioneer contributions in isolation and characterization of large number of helicases from human cells while he was at ICGEB Trieste and published several papers in high impact journals including EMBO J. and Nucleic Acids Research. From India he has cloned the first plant helicase (Plant J. 2000) and presented the first direct evidence for a novel role of a pea DNA helicase (PNAS, USA, 2005) in salinity stress tolerance and pea heterotrimeric G-proteins (Plant J. 2007) in salinity and heat stress tolerance. Dr. Tuteja has reported the first direct evidence in plant that PLC functions as an effector for Ga subunit of G-proteins. All the above work has received extensive coverage in many journals, including Nature Biotechnology, and bulletins all over the world. His group has also discovered novel substrate (pea CBL) for pea CIPK (FEBS J. 2006). He has already developed the salinity tolerant tobacco and rice plants without affecting yield. Recently, few new high salinity stress tolerant genes (e.g. Lectin receptor like kinase, Chlorophyll a/b binding protein and Ribosomal L30E) have been isolated from Pisum sativum and have been shown to confer high salinity stress tolerance in bacteria and plant (Glycoconjugate J. 2010; Plant Signal. Behav. 2010). Recently, very high salinity stress tolerant genes from fungus Piriformospora indica have been isolated and their functional validation in fungus and plants is in progress. Overall, Dr. Tuteja's research uncovers three new pathways to plant abiotic stress tolerance. His results are an important success and indicate the potential for improving crop production at sub-optimal conditions.



Sarvajeet Singh Gill was born on January 21st, 1979. Dr. Gill obtained his B.Sc. (1998) from Y.D. College, Kanpur University and M.Sc. (2001, Gold Medalist), M. Phil. (2003) and Ph.D (2009) from Aligarh Muslim University. Presently, Dr. Gill is working as Assistant Professor in Centre for Biotechnology, MD University, Rohtak, Haryana.

Dr. Gill's main area of research includes Genetic Engineering, Stress Physiology and Molecular Biology (Development of abiotic stress tolerant crop plants,

the physiological, biochemical and molecular characterization of agronomically important plants under abiotic stress factors, involvement of mineral nutrients and other biotechnological approaches in the amelioration of abiotic stress effects in crop plants, use of a combination of genetic, biochemical, genomic and proteomic approaches to understand the responses of various components of antioxidant machinery to abiotic stress and stress signaling and stress tolerance in crop plants. Dr. Gill has several research papers, review articles and book chapters to his credit in the journals of national and international repute and in edited books. He has edited four books namely Sulfur assimilation and Abiotic Stress in Plants; Eutrophication: causes, consequences and control; Plant Responses to Abiotic Stress, Omics and Abiotic Stress Tolerance and Improving Crop Resistance to Abiotic Stress, published by Springer-Verlag (Germany), IK International, New Delhi, Bentham Science Publishers and Wiley-VCH, Verlag GmbH & Co. Weinheim, Germany, respectively. Dr. Gill is a regular reviewer of National and International journals and grants. He was awarded Junior Scientist of the year award by National Environmental Science Academy New Delhi in 2008. With Dr. Tuteja, Dr. Gill is working on heterotrimeric G proteins and plant DNA helicases to uncover the abiotic stress tolerance mechanism in rice. The transgenic plants overexpressing heterotrimeric G proteins and plant DNA helicases may be important for improving crop production at sub-optimal conditions.

List of Contributors

Altaf Ahmad Molecular Ecology Laboratory, Department of Botany, Faculty of Science, Jamia Hamdard (Hamdard University), New Delhi 110062, India e-mail: aahmed@jamiahamdard.ac.in

Aqil Ahmad Plant Physiology Section, Department of Botany, Aligarh Muslim University, Aligarh 202002, UP, India

Faheem Ahmed School of Nano and Advanced Materials Engineering, Changwon National University, Changwon, Gyeongnam 641-773, Republic of Korea

Naser Aziz Anjum Centre for Environmental and Marine Studies (CESAM) & Department of Chemistry, University of Aveiro, Aveiro, Portugal

Nishat Arshi School of Nano and Advanced Materials Engineering, Changwon National University, Changwon, Gyeongnam 641-773, Republic of Korea

Don Aruna Remote sensing Applications Laboratory, School of Environmental Sciences, Jawaharlal Nehru University, New Mehrauli Road, New Delhi 110067, India

Jayasekara Balasuriya Remote sensing Applications Laboratory, School of Environmental Sciences, Jawaharlal Nehru University, New Mehrauli Road, New Delhi 110067, India

Alma Balestrazzi Dipartimento di Genetica e Microbiologia, Università degli Studi di Pavia, via Ferrata 1, 27100 Pavia, Italy e-mail: alma.balestrazzi@unipv.it

Daniela Carbonera Dipartimento di Genetica e Microbiologia, Università degli Studi di Pavia, via Ferrata 1, 27100 Pavia, Italy

Rakhi Chaturvedi Department of Biotechnology, Indian Institute of Technology Guwahati, Guwahati 781039, Assam, India e-mail: rakhi_chaturvedi@yahoo.co.uk; rakhi_chaturvedi@iitg.ernet.in **Massimo Confalonieri** Centro di Ricerca per le Produzioni Foraggere e Lattiero-Casearie (C.R.A.), Lodi, Italy

T. B. Dakua School of Environmental Sciences, Jawaharlal Nehru University, New Mehrauli Road, New Delhi 110067, India

Hema Diwan Molecular Ecology Laboratory, Department of Botany, Faculty of Science, Jamia Hamdard (Hamdard University), New Delhi 110062, India

Mattia Donà Dipartimento di Genetica e Microbiologia, Università degli Studi di Pavia, via Ferrata 1, 27100 Pavia, Italy

Wim Van den Ende Laboratory for Molecular Plant Physiology, Katholieke Universiteit Leuven, Kasteelpark Arenberg 31 box 2434, 3001 Heverlee, Belgium e-mail: Wim.VanDenEnde@bio.kuleuven.be

Ritu Gill Centre for Biotechnology, Maharshi Dayanand University, Rohtak, Haryana, India

Sarvajeet Singh Gill Plant Molecular Biology Group, International Centre for Genetic Engineering and Biotechnology (ICGEB), Aruna Asaf Ali Marg, New Delhi, India

e-mail: ssgill14@yahoo.co.in

Gregory Graff Agricultural and Resource Economics, Colorado State University, B306 Clark Hall, Fort Collins, CO 80523, USA e-mail: Gregory.Graff@colostate.edu

Meetu Gupta Department of Biotechnology, Jamia Milia Islamia, Jamia Nagar, New Delhi 110025, India

Mirza Hasanuzzaman Department of Applied Biological Science, Kagawa University, Miki-cho, Kita-gun, Kagawa, Japan

Shamsul Hayat Plant Physiology Section, Department of Botany, Aligarh Muslim University, Aligarh 202002, UP, India

Rashmi Rekha Hazarika Department of Biotechnology, Indian Institute of Technology Guwahati, Guwahati 781039, Assam, India

Gal Hochman Agricultural, Food, and Resource Economics, Rutgers University, New Brunswick, NJ, USA

Mohd Irfan Plant Physiology Section, Department of Botany, Aligarh Muslim University, Aligarh 202002, UP, India

Shahinul Islam Institute of Biological Sciences, University of Rajshahi, Rajshahi–6205, Bangladesh e-mail: shahin ibsc@ru.ac.bd

Bremont Juan Francisco Jimenez Instituto Potosino de Investigación Científica y Tecnológica, San Luis Potosí, México

Ishrat Khan Molecular Ecology Laboratory, Department of Botany, Faculty of Science, Jamia Hamdard (Hamdard University), New Delhi 110062, India

Bon Heun Koo School of Nano and Advanced Materials Engineering, Changwon National University, Changwon, Gyeongnam 641-773, Republic of Korea e-mail: bhkoo@changwon.ac.kr

D. Sakthi Kumar Bio-Nano Electronics Research Center, Graduate School of Interdisciplinary New Science, Toyo University, Saitama-350-8585, Kawagoe, Japan

e-mail: sakthi@toyo.jp

Shalendra Kumar School of Nano and Advanced Materials Engineering, Changwon National University, Changwon, Gyeongnam 641-773, Republic of Korea

Pradeep Kumara Remote sensing Applications Laboratory, School of Environmental Sciences, Jawaharlal Nehru University, New Mehrauli Road, New Delhi 110067, India

Anca Macovei Dipartimento di Genetica e Microbiologia, Università degli Studi di Pavia, via Ferrata 1, 27100 Pavia, Italy

Santiago Javier Maiale Instituto de Investigaciones Biotecnológicas, Instituto Tecnológico de Chascomús, Consejo Nacional de Investigaciones Científicas y Técnicas, Universidad de Buenos Aires, Av. Intendente Marino Km 8,200 CC 164, (B7130IWA) Chascomús, Prov. de Buenos Aires, Argentina

Ana Bernardina Menéndez Biotechnological Unit 3, Instituto de Investigaciones Biotecnológicas, Instituto Tecnológico de Chascomús, Consejo Nacional de Investigaciones Científicas y Técnicas, Universidad de Buenos Aires, Av. Intendente Marino Km 8,200 CC 164, (B7130IWA) Chascomús, Prov. de Buenos Aires, Argentina

Vijay Kumar Mishra Department of Biotechnology, Indian Institute of Technology Guwahati, Guwahati 781039, Assam, India

Sudip Mitra School of Environmental Sciences, Jawaharlal Nehru University, New Mehrauli Road, New Delhi 110067, India e-mail: sudipmitra@yahoo.com

Saumitra Mukherjee Remote sensing Applications Laboratory, School of Environmental Sciences, Jawaharlal Nehru University, New Mehrauli Road, New Delhi 110067, India

e-mail: saumitramukherjee3@gmail.com

Remya Nair Bio-Nano Electronics Research Center, Graduate School of Interdisciplinary New Science, Toyo University, Saitama-350-8585, Kawagoe, Japan

Darin Peshev Laboratory for Molecular Plant Physiology, Katholieke Universiteit Leuven, Kasteelpark Arenberg 31 box 2434, 3001 Heverlee, Belgium

P. Ragavan Andaman and Nicobar Islands Forests and Plantation Corporation Limited (ANIFPDCL), Port Blair, India

L. Rangan Department of Biotechnology, IIT Guwahati, Assam, India

Robert J. Reid School of Earth and Environmental Sciences, University of Adelaide, Adelaide 5005, Australia e-mail: robert.reid@adelaide.edu.au

Andrés Alberto Rodriguez Instituto de Investigaciones Biotecnológicas, Instituto Tecnológico de Chascomús, Consejo Nacional de Investigaciones Científicas y Técnicas, Universidad de Buenos Aires, Av. Intendente Marino Km 8,200 CC 164, (B7130IWA) Chascomús, Prov. de Buenos Aires, Argentina

Kessler Margarita Rodriguez Instituto Potosino de Investigación Científica y Tecnológica, San Luis Potosí, México

Oscar Adolfo Ruiz Instituto de Investigaciones Biotecnológicas, Instituto Tecnológico de Chascomús, Consejo Nacional de Investigaciones Científicas y Técnicas, Universidad de Buenos Aires, Av. Intendente Marino Km 8,200 CC 164, (B7130IWA) Chascomús, Prov. de Buenos Aires, Argentina e-mail: ruiz@intech.gov.ar; oruiz@iib.unsam.edu.ar

Mani Saxena Andaman and Nicobar Islands Forests and Plantation Corporation Limited (ANIFPDCL), Port Blair, India

Alok Saxena Department of Environment and Forest, Andaman and Nicobar Islands, Port Blair, India e-mail: dr.aloksaxena@gmail.com

Pankaj Sharma Centre for Biotechnology, Maharshi Dayanand University, Rohtak, Haryana, India

C. Kumar Singh Remote Sensing Applications Laboratory, School of Environmental Sciences, Jawaharlal Nehru University, New Mehrauli Road, New Delhi 110067, India

Rakesh Kumar Singh Department of Mycology & Plant Pathology, Institute of Agricultural Sciences, Banaras Hindu University, Varanasi, UP, India e-mail: rakesh_bhuin@hotmail.com

Alok Krishna Sinha National Institute of Plant Genome Research, Aruna Asaf Ali Road, New Delhi 110067, India e-mail: alok@nipgr.ac.in

Amit Srivastava School of Biotechnology, Faculty of Science, Banaras Hindu University, Varanasi, UP, India

Dipesh Kumar Trivedi Plant Molecular Biology, International Centre for Genetic Engineering and Biotechnology (ICGEB), New Delhi, India

Narenda Tuteja Plant Molecular Biology Group, International Center for Genetic Engineering and Biotechnology (ICGEB), Aruna Asaf Ali Marg, New Delhi 110067, India e-mail: narendra@icgeb.res.in

Arif Shafi Wani Plant Physiology Section, Department of Botany, Aligarh Muslim University, Aligarh 202002, UP, India

Dhammaprakash Pandahri Wankhede National Institute of Plant Genome Research, Department of Biotechnology, Government of India, New Delhi, India

David Zilberman Agricultural and Resource Economics, University of California, Berkeley, CA, USA

Chapter 8 In Vitro Haploid Production—A Fast and Reliable Approach for Crop Improvement

Rashmi Rekha Hazarika, Vijay Kumar Mishra and Rakhi Chaturvedi

1 Introduction

1.1 Overview of Haploidy

Haploids have attracted great interest of plant physiologists, embryologists, geneticists and breeders ever since the discovery of the first natural haploid embryos and plants in *Datura stramonium* in 1922 by Blakeslee et al. Subsequently, haploidy has been reported in many species, but at low and variable frequencies and was regarded as a special biological phenomenon. The low frequency of spontaneously arising haploid plants severely limited the utilization of haploids for crop improvement and genetic studies. The remarkable discovery that haploid embryos and plants can be produced through in vitro culture of anthers of *Datura* (Guha and Maheshwari 1964, 1966) brought renewed interest to haploidy. This method of androgenic haploid production was quickly attempted in many species to hasten the breeding programme in several economically-important plants.

The life cycle of angiosperms (higher plants) is characterized by alternating generations of sporophytes and gametophytes. The gametophytic phase arises when the diploid cells undergo meiosis (reduction division) to form male and female gametes. This phase is shortlived as fertilization of the egg re-establishes the diploid sporophytic phase. The sporophytic phase characterized by diploid (2n) chromosome number is the product of fertilization of male and female gametes, containing the haploid (n) set of chromosomes from each parent (Forster et al. 2007). Therefore, haploid is a generalized term for plants that contain the gametic chromosome number (n). This is in contrast to diploid plants, which contain two sets (2n) of chromosomes. Haploids are sexually sterile and, therefore, doubling of the chromosomes is

R. Chaturvedi (🖂) · R. R. Hazarika · V. K. Mishra

Department of Biotechnology, Indian Institute of Technology Guwahati, Guwahati 781039, Assam, India

e-mail: rakhi_chaturvedi@yahoo.co.uk; rakhi_chaturvedi@iitg.ernet.in

required to produce fertile plants which are called doubled haploids or homozygous diploids.

Spontaneous production of haploids usually occurs through the process of parthenogenesis (embryo development from an unfertilized egg). However, occasionally, they bear the characters of male parent only, suggesting their origin through 'ovule androgenesis' (embryo development inside the ovule by the activity of the male nucleus alone where elimination or inactivation of egg nucleus occurs before fertilization) (Bhojwani and Razdan 1996). Although in vivo occurrence of haploids has been reported in several species, the frequency is very low (Pochard and Dumas de Vaulx 1971; Lacadena 1974). Several methods have been employed for producing haploid plants. In 1970, Kasha and Kao reported haploid production in barley following wide hybridization and the subsequent preferential elimination of the chromosomes of wild species during early embryogenesis. This provided a system that produced a large number of haploids across most genotypes (Choo et al. 1985). However, this program has been restricted to only a few crops so far. In vitro haploid plants can be obtained by triggering the male or female gametic cells to undergo sporophytic development. In vitro androgenesis (anther-microspore culture) is one of the most preferred techniques for obtaining haploids, but in vitro gynogenesis (unfertilized ovary-ovule culture) can prove to be a complementary technique in species where anther culture is inaccessible or less productive. It means that not only the microspore, but also the megaspore of angiosperms can be triggered in vitro to undergo sporophytic development. However, production of haploids via gynogenesis is more tedious, less efficient in comparison to androgenesis. In addition to the above techniques, in situ parthenogenesis (pollen irradiation and chemical treatment) can be employed for generation of haploid plants (Kurtar and Balkaya 2010).

Haploid and diploid lines play a vital role in genomics and have been used for the purpose of physical mapping, genetic mapping and also for integration of genetic and physical mapping. Additionally, haploid and doubled haploid plants are adapted for mutagenesis and genetic transformation experiments, presenting the advantage of immediate production of homozygous lines. It is also expected that, in the near future, haploid and doubled haploid plants will play an increasingly important role in whole genome sequencing projects, where homozygosity is of particular advantage.

1.2 History of Haploids

Dorothy Bergner in 1921 was the first to describe the natural occurrence of sporophytic haploids in the weed species *Datura stramonium* and this was reported by Blakeslee et al. in 1922. This report was followed by similar reports in several other crop species. The initial attempts to use haploidy in breeding were started by Chase in 1952 who selected the low frequency of parthenogenetic haploids in maize and then applied chromosome doubling treatments to produce inbred lines (Kasha and Maluszynski 2003).

Ever since the discovery of the first haploid in *Datura stramonium* in 1921, several researchers attempted to induce the unfertilized egg cell or other cells of the embryo sac to undergo parthenogenesis via in vivo means. In vivo induction of haploidy is carried out by the application of various physical, chemical or biological stimulants (Yang and Zhou 1982). This was followed by repeated attempts by various scientists to improve the frequency of parthenogenesis. While ovule and ovary culture suffered setbacks, Guha and Maheshwari (1964) made a breakthrough in anther culture (Yang and Zhou 1982). Pioneering work by Guha and Maheshwari in 1964 and 1966 opened a new vista for haploid breeding through in vitro means. It was a major advancement in haploid breeding of higher plants in which development of numerous pollen plantlets through in vitro anther culture of Datura innoxia was achieved. They wanted to study normal breeding development, but they found development of embryos in their cultured anthers (Maheshwari et al. 1980, 1982). These embryos developed into plants with a haploid chromosome number. However, the frequency of haploid production was significantly low. These studies led to further experiments to improve the frequency of haploids through in vitro anther culture. Plant regeneration via anther culture has been reported in more than 250 plants (Maluszynski et al. 2003a).

Consequently, attention shifted mainly to anther culture and culture of female tissues was neglected for about a decade. However, other researchers continued to work in the field of in vitro gynogenesis and in 1971, Uchimiya et al. observed the division of haploid cells in callus tissues obtained from cultured unpollinated ovaries of *Zea mays* and ovules of *Solanum melongena*. But this came into lime-light only in 1976 when San Noeum reported development of haploid plantlets in unfertilized ovary cultures of *Hordeum vulgare*. Subsequently, Zhu and Wu (1979) obtained haploid plants from cultured unfertilized ovaries of *Triticum aestivum* and *Nicotiana tabacum*. This was followed by development of haploid plantlets via unfertilized ovary culture in several economically-important plant species such as *Zea mays* (Tang et al. 2006), *Psoralea corylifolia* (Chand and Sahrawat 2007), *Cucurbita pepo* (Shalaby 2007), *Guizotia abyssinica* (Bhat and Murthy 2007), *Cocos nucifera* (Perera et al. 2007), *Morus alba* (Thomas et al. 1999), etc.

1.3 Current Status

There are a number of excellent reviews on the production of haploids and doubled haploids so far, including those of Andersen (2005), Dunwell (2010), Germanà (1997, 2006, 2007, 2009), Kasha (1974), Magoon and Khanna (1963), Maluzynski et al. (2003a, 2003b), Palmer et al. (2005), Touraev et al. (2009), Zhang et al. (1990) and Xu et al. (2007). The doubled haploid techniques have been well established in a range of economically-important crop species, including major cereals and cabbage (Wedzony et al. 2009). Regeneration via anther culture has been reported in more than 250 species belonging to Solanaceae, Cruciferae and Gramineae families (Dunwell 1986; Germanà 2011; Hu and Yang 1986), while there are limited reports

on many legumes and woody plants as they are rather recalcitrant (Sangwan-Norreel et al. 1986; Bajaj 1990; Raghavan 1990; Wenzel et al. 1995; Germanà 2006, 2009, 2011). Recent advances in anther culture have been reported by Dunwell (2010), Wedzony et al. (2009), Pratap et al. (2009), Srivastava and Chaturvedi (2008); and Touraev et al. (2009). Unfertilized ovary–ovule culture has been applied to species including sugar beet, tulip, cucumber, sweet potato, onion, squash, gerbera, rice, maize, niger and tea (Chen et al. 2011). Wide hybridization method for production of haploids is routinely used in wheat and other cereal breeding programmes. The general method followed involves a phase of embryo rescue in vitro, usually followed by chromosome doubling with colchicine (Maluszynski et al. 2003a).

1.4 Efficiency of Haploid Production

Homozygous lines are of utmost importance in breeding programmes. Wide hybridization and anther/microspore culture are two of the most preferred techniques for doubled haploid production in crop plants. There is not much difference in the cost of haploid production or in the time and amount of labour required to produce haploids via these means. Currently, gynogenesis is the least favoured technique because of the low efficiency of production of haploids, but the value of doubled haploids in species that do not respond to other methods of haploid production makes this method worthwhile (Maluszynski et al. 2003a; Touraev et al. 2001). The limitation of wide hybridization is that it is restricted to the cereals where the chromosome elimination system appears to operate. Each crop has different requirements and, thus, there is need for extensive research to develop an efficient system. Wide hybridization has the advantage of being quite effective across genotypes and produces little or no induced variation from cultured embryos. However, anther/microspore culture scores higher over distant hybridization due to the fact that anthers harbour large numbers of haploid microspores per anther. In barley and wheat, where both systems have been comparatively well developed, the yield of green plants from isolated microspore culture can be up to 100 times higher than from wide hybridization in the most responsive genotypes (Kasha and Maluszynski 2003).

Anther culture is feasible in most species but it generally takes plenty of time to develop a competent system in some crops. Moreover, a good aseptic technique is a necessity even though the available methods are simple and reproducible (Maluszynski et al. 2003a). In general, haploid plants are regenerated in vitro from the microspores contained in the anther and require chromosome doubling treatments using chemicals such as colchicine, pronamide, trifluralin, oryzalin and amiprophos methyl (APM) (Wan et al. 1991). A few species such as barley regenerate a large number of doubled haploids as a result of induced chromosome doubling during early division of the microspores (Kasha 2005). Although the application of anther culture is widespread, the presence of the sporophytic anther wall serves as

a hindrance towards complete access of the microspores. There is a need to ensure direct embryogenesis to eliminate an intermediary callus phase that can promote gametoclonal variation among regenerants. The large number of successful reports on haploid production through anther culture stands as a witness of the immense benefits that this system offers. However, the presence of extraneous tissue in the form of anther wall makes this system complex for genetic studies where precision is inevitable.

As haploid plants in crop species obtained from microspore cultures through embryogenesis rather than from callus, the problem of extensive culture induced variability found in earlier reports on anther culture can be significantly done away with. With improvement of frequencies from microspores, the genotypic competence has also been reduced so that the system can be used in breeding programs. The feature of a high frequency of spontaneous chromosome doubling in some crops which results in completely fertile doubled haploid plants without subsequent doubling treatments, is another advantage of microspore culture. In breeding, the instant production of true breeding lines in diploid or allopolyploid species saves a number of generations in the breeding program. High density cultures of synchronized microspores can be set up containing more than thousand embryos per ml of culture media. Although a newer technique in comparison to anther culture, the potential of microspore culture has been realized quite early and efficient procedures have been successfully developed to produce doubled haploids in microspore cultures of tobacco, rapeseed, pepper, wheat, barley, rice, etc (Maluszynski et al. 2003a; Touraev et al. 2001).

2 Methodologies of In Vitro Haploid Production

2.1 Androgenesis

In androgenesis, immature pollen grains are induced to follow the sporophytic mode of development with the application of various physical and chemical stimuli. There are two methods for in vitro production of androgenic haploids, viz anther culture and pollen culture.

2.1.1 Anther Culture

Anther culture imparts an easy and one step protocol for haploid plant production (Chaturvedi et al. 2003). There is no single specific condition or protocol for inducing in vitro androgenesis in all plants since different species and even different cultivars within a species show diverse requirements. The following methods were adopted by Chaturvedi et al. (2003) to initiate in vitro androgenesis in neem plants (*Azadirachta indica* A. Juss.):



Fig. 8.1 a Neem flower buds of 2 mm size bearing correct stage of microspores (6X) b An uninucleate microspore of neem stained with acetocarmine (2300X)

Plant Material

For anther culture, 2 mm size flower buds (Fig. 8.1a) were collected early in the morning during the flowering season (April-May). The stage of microspore development was determined by acetocarmine squash preparations. Anthers containing early-to-late uninucleate stage of microspores (Fig. 8.1b) were cultured in the laboratory.

Establishment of Aseptic Cultures

The flower buds, collected from a field, were rinsed with sterile distilled water (SDW) several times, followed by surface sterilization in a glass vial with 0.1 % mercuric chloride solution (HgCl₂) for 7 min. Finally, the buds were washed three to four times with SDW. The buds were dissected out with the aid of a binocular microscope using pre-sterilized Petriplates, forceps and fine needles. Damaged anthers, if any, were discarded. Twenty anthers, from two buds, were cultured in 55×15 mm pre-sterilized, disposable Petriplates containing 10 ml of MS (Murashige and Skoog 1962) medium. The Petriplates were sealed with parafilm and subjected to specific treatment conditions.

Culture Media Preparation

A number of media compositions containing different concentrations of auxins and cytokinins were used for inducing in vitro androgenesis in neem. MS media, devoid of any growth regulator, did not induce any morphogenic response in anther cultures. For successful induction of callus from anthers, a combination of auxins and a cytokinin was found to be necessary in most of the cultures.

Analytical grade chemicals were used for preparation of stock solution and cultured media. Stock solution for MS/N_6 media preparation can be summarized as follows:

- Macronutrient 10X; Micronutrient 20X; Iron stock 20X; Vitamins 20X
- Growth regulator is used at a concentration of 1×10^{-3} M

All constituents were stored at 4 °C for further use. Myo-inositol and sucrose were added freshly at the time of media preparation. The pH of the medium was adjusted at 5.8 with 1N HCl or 1N NaOH. The media were solidified with 0.8 % agar and autoclaved for 15 min at 1.06 kg cm⁻² and 121 °C, before pouring in 60 mm size Petriplates.

Culture Conditions

The anthers on each media were subjected to an array of pre-treatment conditions, like cold and heat shock for various duration under both dark and light incubation. Light and dark incubation at 25 °C temperature served as a control in all the experiments. All the cultures were maintained continuously at 25 ± 2 °C and 50–60 % relative humidity under 16/8 h (light/dark) photoperiod with diffuse light (1,000–2,000 lx) provided by cool daylight fluorescent tubes (Philips TL40 W). Anther cultures were initially kept continuously in the dark, but after eight weeks, the calli that had developed from these cultures were transferred to multiplication medium and maintained in light.

Twentyfour cultures were raised for each treatment, and each experiment was repeated at least three times. The cultures were observed periodically and the morphological changes were recorded at weekly intervals.

2.1.2 Pollen/Microspore Culture

Anther culture is beset with a number of problems. One of the major problems of anther culture is the concomitant callusing of anther wall along with the pollen, as a result of which the final tissue derived may not be of purely gametophytic origin. Moreover, the plants arising from an anther would constitute a heterogenous population. It has been observed in some species that anther cultures show asynchronous pollen development, the older grains may suppress the androgenic potential of younger grains by releasing toxic substances (Bhojwani and Razdan 1996). Kameya and Hinata (1970) reported for the first time callus formation in isolated pollen cultures of *Brassica oleracea* and the hybrid *B. oleracea* X *B. alboglabra* following which successful pollen-derived androgenic plants have been produced in many crop plants. There are numerous advantages of microspore culture over anther culture for haploid plant production. Unlike anther culture, isolated microspore culture allows haploid plant regeneration directly from microspores, assuring pure gametophytic origin and, thus, avoiding mixing of proliferating anther walls. A homogeneous prep-

aration of pollen at the developmental stage, most suitable for androgenesis, can be obtained by gradient centrifugation. Isolated pollen can be modified genetically by mutagenesis or genetic engineering before the culture and a new genotype can be selected at an early stage of development. Some of the important factors that affect induction of androgenesis in cultured pollen are the composition of the culture medium, pre-treatment and plating density. In most of the cereals, pollen culture involves pre-culture of the anthers for a few days or co-culture of pollen with a nurse tissue. Treatment of pollen-derived embryos and pollen derived callus to recover complete plantlets is similar to that of anther culture. However, the nutritional requirements of isolated pollen in culture are more complex than those of cultured anthers.

2.2 Gynogenesis

As mentioned earlier, in vitro gynogenesis is used as a complementary technique in species where anther/pollen culture is inaccessible or unsuccessful. Young flowers, ovaries or ovules have been used as explants to produce gynogenic haploids where the plants develop from unfertilized cells of female gametophyte (embryo sac) either through direct embryogenesis or via callusing. The following techniques are generally used for production of haploids via in vitro gynogenesis.

2.2.1 In situ Parthenogenesis Induced by Irradiated Pollen Followed by In Vitro Embryo Culture

Parthenogenesis induced in vivo by irradiated pollen followed by in vitro culture of embryos can be an alternative method of obtaining haploids in fruit crops. Gynogenesis by *in situ* pollination with irradiated pollen has been successfully used for *Malus domestica* (L.) Borkh (Zhang and Lespinasse 1991), *Pyrus communis* L. (Bouvier et al. 1993), *Actinidia deliciosa* (A. Chev) (Pandey et al. 1990; Chalak and Legave 1997). This method is based on in vitro culture of immature seeds or embryos, obtained as a result of pollination by irradiated pollen with gamma rays from cobalt 60. The method is useful in those species in which in vitro anther culture has not been successful. Irradiation does not hinder pollen germination but prevents pollen fertilization, thus stimulating the development of haploid embryos from ovules. The success of this technique is dependent on the choice of radiation dose, the developmental stage of the embryos at the time of culture, culture conditions and media requirements.

2.2.2 Ovary Slice Culture

Ovary slice culture technique involves culture of transverse sections of unpollinated ovaries on culture media. The various steps of ovary slice culture of tea are sum-

marized below. The following methods were used to induce in vitro gynogenesis in tea plants (Hazarika and Chaturvedi; unpublished data).

Plant Material

For ovary slice culture in tea, unopened and unpollinated mature flower buds (6-10 mm) size were collected early in the morning. Some of the buds were fixed in FAA (5:5:90 v/v/v Formaldehyde: Acetic acid: 70 % Ethanol) for 48 h and then stored in 70 % alcohol. Later, the appropriate developmental stage of the embryo sac was determined by histological analysis.

Establishment of Aseptic Cultures

The mature unpollinated tea flower buds were surface sterilized with 0.1 % $HgCl_2$ for seven minutes, followed by rinsing with sterile distilled water at least thrice. Carefully dissected transverse sections of ovaries were cultured on Murashige and Skoog's (MS) media supplemented with varying concentrations of auxins and cytokinins. Six ovary slices containing unpollinated ovules were cultured in 60×15 mm pre-sterilized disposable Petriplates containing 10 ml MS medium. The sealed Petriplates were subjected to various regimes of temperature and light treatments.

Culture Media Preparation

Murashige and Skoog (1962) basal medium was used throughout the studies to raise ovary slice cultures. MS basal medium was supplemented with a range of growth regulators, viz., 2, 4-Dichlorophenoxyacetic acid (2, 4-D), α -Naphthalene acetic acid (NAA), N₆- Furfuryladenine (Kinetin), 6-Benzylaminopurine (BAP) and Thidiazuron (TDZ) either individually or in combinations. The media contained 3 % sucrose and was gelled with 0.8 % Agar.

Culture Conditions

The ovary slice cultures from each media were subjected to an array of pre-treatments, like cold (4 °C) and heat shock (33 °C), for various durations under both light and dark incubation. The light and dark incubation at 25 °C served as a control in all the experiments. After the application of treatments, the cultures were maintained continuously at 25 ± 2 °C condition with 50–60 % relative humidity under 16/8 h (light/dark) photoperiod irradiance (1000–2000 lx) provided by cool daylight fluorescent tubes. The cultures were observed periodically and the morphological changes were recorded at weekly intervals.

2.2.3 Ovule Culture

Embryos are difficult to excise. To prevent damaging the embryos during the excision process, they are sometimes cultured while still inside the ovule. This technique is referred to as ovule culture or in ovulo embryo culture (Reed 2005). The unfertilized ovary is surface sterilized and the ovules are taken and placed into the culture. Excision of ovule, followed by culture on specific media may be either extremely easy to accomplish, as in case of large-seeded species in which only a single ovule is present, or time-consuming and intricate, as in small-seeded polyovulate species. Two types of ovule support systems have been developed. The filter paper support system involves culturing of the ovules on top of filter paper placed over liquid medium, whereas the vermiculite support technique demands placing the ovules on a sterile vermiculite/liquid media mixture (vermiculite support) with the micropylar side down. Unpollinated ovule culture has been used for haploid production in sugar beets and onions. Since there is usually only one egg cell per ovule, ovule culture has much less potential than microspore culture (Kasha and Maluszynski 2003). In Nicotiana rustica cv Rustica ovules with placenta were isolated from flower buds and were cultured on N6 medium supplemented with growth regulators (Katoh and Iwai 1993).

2.3 Wide Hybridization

Haploids can also be induced by a process of selective chromosome elimination that follows certain interspecific pollinations. This phenomenon was discovered first in barley with crossing between Hordeum vulgare and H. bulbosum (Kasha and Kao 1970) and is now used routinely in wheat and other cereal breeding programmes; haploids were induced in these species following pollination with maize pollen. High frequency gynogenic haploids of Triticum aestivum have been raised by crossing them with H. bulbosum followed by embryo culture (Barclay 1975; Zenketler and straub 1979; Inagaki 1990). The process involves a phase of embryo rescue in vitro, usually followed by chromosome doubling with colchicine. The term "embryo rescue" refers to an vitro techniques the purpose of which is to promote the development of an immature or weak embryo into a viable plant (Reed 2005). Embryo rescue has been widely used to raise plants from hybridization in which failure of endosperm development causes embryo abortion. In embryo rescue procedures, the artificial nutrient medium serves as a substitute for the endosperm, thereby allowing the embryo to continue its development. Embryo rescue techniques are among the oldest and most successful in vitro procedures. One of the primary uses of embryo rescue has been to produce interspecific and intergeneric hybrids. While interspecific incompatibility can occur for a wide variety of reasons, one common cause is embryo abortion. The production of small, shrunken seed following wide hybridization is indicative of a cross in which fertilization occurs but seed development is disrupted. Embryo rescue procedures have been very successful in overcoming this barrier among wide hybridizations in a wide range of plant species (Collins and Grosser 1984). In addition, embryo rescue has been used to recover maternal haploids that have developed as a result of chromosome elimination following interspecific hybridization.

3 Factors Affecting Haploid Production

There are numerous endogeneous and exogeneous factors that affect in vitro haploid production. These factors can be genetic, physiological, physical and chemical may also interact amongst each other to divert the microspores/egg cells to enter into a new developmental pathway. Some of the crucial factors affecting haploid production in plants are discussed below:

3.1 Genotype of the Donor Plant

The genotype of the donor plants has great influence on anther culture response and has been known since the early days of development of plants from pollen grains (Nitsch and Nitsch 1969; Maheshwari et al. 1980). However, it is only in recent times that the genetic factors affecting androgenesis have been studied more intensively. Many crop species are quite recalcitrant in their in vitro response which is governed by specific genes on the chromosomes (Datta 2005). Certain regions of the chromosomes appear to be associated with the formation of embryo like structures (Wan et al. 1992). As has been mentioned by Zhou in 1996, additive gene effects explain most of the variations observed across diverse genotypes, but cytoplasmic influences and non additive gene effects also play important roles in determining in vitro regeneration ability of anther-derived cultures. In earlier studies, significant difference in callus formation using varieties or crosses was observed. In some species, only a few genotypes have responded of the many that were tested. In fact, genetic factors contribute in a major way to the differences in the number of haploid plants produced (Custodio et al. 2005; Sopory and Munshi 1996).

As in anther culture, a difference in response also exists among donor cultivars in ovary and ovule culture. It was reported by Zhu et al. (1981) that the percentage of ovaries producing gynogenic calli in four wheat cultivars varied from 1.3 to 10.9 %. In *Nicotiana tabacum*, two cultivars had an induction frequency as high as 75 % and 80 %, but in another species, *N. rustica*, it was only 8 % (Wu and Chen 1982). Similar genotypic competence for gynogenesis has been reported in a number of crop species (Keller 1990; Lux et al. 1990; Tosca et al. 1999; Kobayashi et al. 1993; Sibi et al. 2001; Alan et al. 2004). It has been observed that genotypic competence exists even between different cultivar types within the same plant species, as has been reported in *Guizotia abyssinica* (Bhat and Murthy 2007).

3.2 Physiological Status of the Donor Plant

The physiological conditions of the donor plant, i.e., the environmental conditions and age of the donor plant, directly affect both in vitro androgenesis and in vitro gynogenesis in almost all plant species. A correlation between plant age and anther response has also been demonstrated by various scientists. Generally, the first flush of flowers vields more responsive anthers than those that are born later. Similar is the case with ovary culture. The frequency of androgenesis is usually higher in anthers harvested at the beginning of the flowering period and showed a gradual decline in relation to plant age (Bhojwani and Razdan 1996). However, it has been reported in Brassica napus and B. rapa that pollen from older, sickly looking plants yielded a greater number of embryos than those from young and healthy plants. Varying temperature and light conditions during the growth of donor plants also affect anther response. In anther culture of grape, the induction frequency of embyoids, derived from spring flowers, was higher than that derived from summer flowers (Zou and Li 1981). The microscopical observations showed that some varieties of rubber trees often have a lot of degenerated and sterile microspores in their anthers, in early spring or hot summer, due to the influence of unfavourable climatic conditions. As a result, no pollen embryoids were obtained from such anthers, but only somatic calli were obtained (Chen et al. 1982).

3.3 Stage of Explants Material at the Time of Inoculation

3.3.1 Stage of Microspores

The stage of microspores at the time of inoculation is one of the most critical factors for induction of androgenesis. Detailed cytological studies conducted on poplar, rubber (Chen 1986) and apple (Zhang et al. 1990) have shown that androgenic callus and embryos were mainly induced through a deviation of the first pollen mitosis to produce two undifferentiated nuclei. Besides affecting the overall response, the stage of microspore at culture also has a direct effect on the ploidy of plants produced in anther culture (Sunderland and Dunwell 1977). About 80 % of the embryos obtained from binucleate microspores of *Datura innoxia*, a highly androgenic species, were non haploids (Sunderland et al. 1974). In a vast majority of species where success has been achieved, anthers were cultured when microspores were at the uninucleate stage of microsporogenesis (Chaturvedi et al. 2003; Pedroso and Pais 1994; Sopory and Munshi 1996).

3.3.2 Stage of the Embryo Sac

It has been reported that the effect of ovule development on gynogenesis is profound as it harbours the embryo sac comprising the egg cell. The stage of embryo sac is an important determining factor for in vitro gynogenesis in various plant species. However, it is difficult to know the stage of embryo sac at the time of inoculation. Several authors prefer to describe the inoculation stage according to the developmental stage of the flower bud or stage of pollen development. However, this could not be possible in several species, where male and female gametophytes do not mature simultaneously, a phenomenon known as protandry, the maturation of anthers before carpels (e.g., onion, leek, sunflower, sugar beet, carrot,) and the opposite protogyny (e.g., pearl millet). In such cases, the stage of embryo-sac at culture can be determined by histological preparations of ovary/ovules that are at identical stage with that of cultured ovary/ovules (Srivastava et al. 2009).

The unpollinated ovules collected two or three days before anthesis were non responsive on culture media as has been reported in Niger whereas unpollinated ovules collected one day before anthesis were most responsive with about 5 to 13.3 % embryogenesis recorded (Bhat and Murthy 2007). Although a wide range of embryo sac stages are responsive to gynogenic development, in most cases, nearly mature embryo-sac stage gave better results. This is quite contrary to anther culture in which mature pollen is nonresponsive to androgenesis. In barley and rice, unfertilized ovary cultures with late staged mature embryo sacs gave good results (San Noeum 1976, 1979; Wang and Kuang 1981) while others reported success with ovary cultures containing uninucleate to mature embryo sacs (Zhou and Yang 1981; Yang and Zhou 1982; Kuo 1982; Huang et al. 1982).

3.4 Culture Media

3.4.1 Basic Media

The constituents of the basal medium and combinations of growth regulators serve as an important factor in eliciting successful androgenesis and gynogenesis. However, it is difficult to suggest one single culture medium with a particular growth regulator for all the systems. Regeneration of androgenic and gynogenic plants may occur directly via embryogenesis or via callus formation from pollen/egg cell, followed by organogenesis. In the later stages of plant development, the media constituents may vary according to culture conditions and requirement by the plant itself.

Most species exhibit androgenesis on a complete nutrient medium (mineral salts, vitamins and sucrose) with or without growth regulators. Most commonly used basal media for anther culture are N_6 (Chu 1978) medium, MS (Murashige and Skoog 1962) medium with slight modifications, Nitsch and Nitsch (1969) medium, and B5 medium (Gamborg et al. 1968). Half strength MS medium is suitable for Solanaceae and N_6 medium has been used for cereals (Chu 1978).

Most early work in 1950s used Nitsch medium for unpollinated ovule and ovary culture; however since the 1970s, Miller (1963), MS or N_6 media have been used in successful experiments. In *Gerbera*, MS seems better than the Knop's and Heller medium (Cagnet-Sitbon 1980). An increase in the content of B group vitamins and glycine in H medium has been reported to have promoted induction-frequency in tobacco ovary culture (Wu and Chen 1982).

3.4.2 Growth Regulators

The requirement of growth regulators and culture medium in terms of kind and concentration may vary with each and every plant system. Generally, there is an agreement that the source and amount of total nitrogen as well as combination of a cytokinin and an auxin is necessary for pollen embryogenesis and pollen callusing in several woody plants (Chaturvedi et al. 2003; Chen 1986; Nair et al. 1983). It has been reported in most members of Solanaceae, that addition of an auxin to the induction medium is not a pre-requisite for anther response, but the addition of auxins and cytokinins alone or in combination is crucial for microspore-derived embryo induction in majority of the plants, especially the recalcitrant ones (Maheshwari et al. 1982). The type and concentration of auxins seem to determine the pathway of microspore development (Ball et al. 1993), with 2, 4-D inducing callus formation, and indole-3-acetic acid (IAA) and Napthaleneacetic acid (NAA) promoting direct embryogenesis (Armstrong et al. 1987; Liang et al. 1987). Gibberellins and abscisic acid have been occasionally added to the media.

Growth regulators, especially, auxins are widely used for induction of gynogenesis and their optimum concentrations have been reported to vary considerably from species to species (San Noeum and Gelebart 1986). In sunflower, gynogenesis occurs only when 2,4-D or NAA is added to the medium (Gelebart and San Noeum 1987). As observed in many species, a combination of auxin and cytokinin was also reported to be useful for gynogenesis in allium species (Alan et al. 2003) and mulberry (Thomas et al. 1999). In mulberry, gynogenic haploids are also produced on BA or Kinetin medium (Lakshmi Sita and Ravindran (1991). These researchers observed gynogenesis in *ab initio* ovary cultures of mulberry. Thomas et al. (1999) obtained maximum gynogenic response when excised ovaries from inflorescence segments of mulberry were cultured on MS supplemented with BAP (8.5 μ M)+2,4-D (4.5 μ M), followed by transferring them to MS+2,4-D (4.5 μ M)+Glycine (6660 μ M)+Proline (1738 μ M).

3.4.3 Growth Additives

The supplement of other substances, such as free amino acids (glutamine, proline, glycine), biotin, myo-inositol, casein hydrolysate, coconut water, silver nitrate (ethylene antagonist) and polyvinylpyrrolidone has been reported to enhance gynogenic response (Reinert and Bajaj 1977; Powell 1990; Achar 2002). Moreover, the addition of exogenous aliphatic polyamines (PAs) to the culture medium has been found to increase the number of pollen-derived embryos in potato (Tiainen 1992), in some Indian wheat cultivars (Rajyalakshmi et al. 1995), cucumber (Ashok kumar et al. 2004 and Chiancone et al. 2006). Polyamines such as putrescine, cadaverine, spermidine and spermine are low molecular mass polycations which are involved in in vitro organogenesis and embryogenesis (Bagni and Tassoni 2001; Kumar et al. 1997). Similarly, the use of additives has also been reported in unfertilized ovary cultures. Thomas et al. (1999) obtained maximum gynogenic response in mulberry when excised unfertilized ovaries from inflorescence segments were transferred from MS+BAP (8.5 μ M)+2,4-D (4.5 μ M) to MS+2,4-D (4.5 μ M) + Glycine (6660 μ M)+Proline (1738 μ M).

3.4.4 Carbon Source

Sucrose has generally been used as the major carbohydrate source in the culture medium. Sucrose concentration in induction medium has a major effect on osmosis, and the development of embryos is apparently influenced by osmosis (Wakizuka and Nakajima 1975). The effect of sucrose on anther culture has been investigated in a number of species. The necessity of sucrose for successful androgenesis was first demonstrated by Nitsch in 1969 for tobacco and later by Sunderland in 1974 for *Datura innoxia*. Generally, sucrose is supplied at 2–3 % concentration. However, increase in its concentration can lead to beneficial morphogenic potential (Agarwal et al. 2006; Sopory and Munshi 1996) by suppressing the proliferation of anther wall (somatic tissues) (Ouvang et al. 1973). For potato, 6 % sucrose proved distinctly superior than 2–4 % sucrose in terms of the number of anthers forming pollen embryos (Sopory et al. 1978). High sucrose levels (6–17%) are required in Gramineae and Brassicaceae families in which mature pollen is shed in the tricellular condition (Dunwell and Thurling 1985), whereas for those in which mature pollen is bicellular (e.g., Solanaceae), lower level of sucrose, such as 2-5 %, is usually beneficial (Dunwell 2010). All Brassica species require 12-13 % sucrose for androgenesis in anther and pollen cultures. According to (Dunwell and Thurling 1985), high sucrose concentration favours better survival of pollen grains, thus improving the frequency of androgenesis in *Brassica napus* (Last and Bretell 1990). Wedzony et al. (2009) have reported the beneficial effects of maltose on androgenesis in anther cultures of wheat, triticale, rye and rice.

The type and concentration of sucrose used in the medium for inducing in vitro gynogenesis varies from species to species. High sucrose concentration (8–10 %) in the culture medium has been shown to be helpful in some species, like sweet potato (Kobayashi et al. 1993) and onion (Campion et al. 1992), whereas in summer squash, 9 % sucrose is detrimental for production of any embryos (Shalaby 2007). Sucrose concentration used in unfertilized ovary and ovule culture was 3 to 10 % in barley, 4 to 8 % in wheat, 3 to 6 % in rice, 2 % in tobacco and 3 to 6 % in *Gerbera* (Yang and Zhou 1982).

3.5 Culture Conditions

Pre-treatments, such as chilling, temperature shock, high humidity, water stress, anaerobic treatment, centrifugation, sucrose, nitrogen starvation, ethanol, gamma radiation, microtubule disruptive agents, electrostimulation, high medium pH, heavy metal treatments are particularly popular approaches in anther and microspore cultures (Shariatpanahi et al. 2006).

Temperature treatment is considered to be the most effective treatment to induce pollen embryogenesis and it may be applied to excised flower buds or whole inflorescences before culture or cultured anthers or pollen grain, prior to their transfer to standard culture room conditions, in order to divert the gametophytic pathway to sporophytic mode of development. The optimum temperature and duration of pre-treatments vary with the genotype. Cold pre-treatment (4 °C for 2–3 days) is employed routinely in the anther culture of many crops and its effect is also genotype-dependant (Osolnik et al. 1993; Powell 1988). In *Brassica* species, a short-duration high temperature pre-treatment at 30–35 °C to cultured anthers before shifting them to 25 °C is required to efficiently switch the developmental pathway.

Light does not seem to be necessary for the induction of androgenesis. For pollen culture of *Datura innoxia* (Sangwan-Norreel 1977), *Nicotiana tabacum* (Sunderland and Roberts 1977) and *Annona squamosa* (Nair et al. 1983), an initial incubation of cultures in dark, followed by diffuse light was found to be suitable. Isolated pollen cultures are more sensitive to light than anther cultures (Nitsch 1977). In *Brassica juncea* (Sharma and Bhojwani 1989) and *Hordeum vulgare* (Xu 1990) species, light is detrimental even for anther cultures. Likewise, neem callusing from microspores requires continuous dark incubation of anthers (Chaturvedi et al. 2003).

A beneficial role of cold treatment on gynogenesis has been reported in some species whilst in others, no significant effects on gynogenesis were observed. Pretreating the capitula of sunflower at 4 °C for 24–48 h before culture, significantly increases the induction frequency (Yan et al. 1987). Cai et al. (1988) observed a promotory effect of cold treatment to young panicles of rice at 7 °C for one day prior to ovary culture in gynogenesis. In *Cucumis melo*, pollination of pistils with irradiated pollen is essential to obtain ovules capable of forming gynogenic haploids (Katoh et al. 1993).

3.6 Other Miscellaneous Factors

3.6.1 Anther Wall Factor

One of the important research subject in anther culture for woody plants is to avoid the over-proliferation of callus from anther wall tissues and to achieve a high yield of pollen embyoids and pollen calli. In anther culture of most woody plants, both pollen calli or embryoids and somatic calli from anther wall tissues grow simultaneously. The development of callus from somatic tissues of anther can be avoided by culture of isolated microspores. However, there are not many successful reports on microspore culture in woody plants (Chaturvedi et al. 2003). Pelletier and Ilami (1972) had introduced the concept of "wall factor", according to which the somatic tissues of anther play an important role in the induction of sporophytic divisions in pollen. In anther culture of rubber, 47 % of the microspores in close contact with the surrounding somatic cells could develop into multicellular masses as compared to only 5 % of microspores away from the wall (Chen 1986). Anther wall callusing is regarded as a pre-requisite for the formation of androgenic haploids (Chaturvedi et al. 2003; Chen et al. 1982; Chen 1986).

3.6.2 Microspore Culture Density

The culture density is a critical factor in isolated pollen culture. Huang et al. (1990) made a detailed study on the effect of culture density on embryogenesis in pollen cultures of *Brassica napus*. According to this report, the minimum density required for embryogenesis is 3000 pollen/ml of the culture medium but highest embryo yield was obtained at 10,000 to 40,000 pollen/ml. This high plating density is crucial only for the initial couple of days. Dilution of the density from 30,000–40,000 to 1,000 pollen after two days of culture did not reduce the embryogenic frequency. Arnison et al. (1990) reported the effect of culture density in anther cultures of *B. oleracea*. The frequency of pollen embryogenesis was enhanced if the anther culture density was increased from three anthers per 4 ml to 12–24 anthers per ml of the medium. Cardy in 1986 reported that in *B. napus*, the response was better when anthers were cultured at a density of two anthers per ml.

3.6.3 Effect of Female Flower Position

Position of female flowers on the plant stem affected induction of embryos from ovule cultures of *Cucurbita pepo* L (Shalaby 2007). One of the possible explanations for enhancing responses of tissue culture could be attributed to indigenous hormonal level (Johansson 1986).

4 Haploid Production in Economically Important Dicotyledonous Species

4.1 Azadirachta indica A. Juss (Family: Meliaceae)

Azadirachta indica A. Juss, commonly referred to as neem, is an adaptable, tropical, evergreen tree of the family Meliaceae. It thrives best in hot, dry climates where shade temperatures often reach 50 °C and annual rainfall ranges from 400 to 1,200 mm. The neem plant is a native of South and Southeast Asia and grows well in tropical and subtropical areas of the world. The plant is well known for its numerous medicinal, agrochemical and economic uses, which can be attributed to the presence of azadirachtin, a highly oxidized limonoid (triterpenoid) found prominently in the seed kernels. It possesses insect repellent, antifeedant, larvicidal,
growth inhibiting properties against a wide range of pests and thus, has been well accepted as an eco-friendly, biodegradable biopesticide. Today, due to the remarkable properties shown by azadirachtin and other related triterpenoids, the tree has attained universal significance. Almost each and every part of this tree, particularly the leaves, bark and seeds, has manifold applications. Besides being a popular avenue tree, with a large crown, the wood of this tree is used as timber for house building, furniture and other domestic and agricultural tools.

Androgenesis is a very useful technique for resolving the problem of self-incompatibility, heterozygosity and long gestation period in this tree species. However, limited attempt has been made for the overall improvement of this valuable tree through in vitro haploid production. Gautam et al. (1993) made the initial attempts to produce haploids of neem for which they cultured anthers at the uninucleate stage of microspores and observed formation of multicellular pollen. However, all the plants regenerated from anther callus were diploids. Chaturvedi et al. (2003), for the first time, generated androgenic haploids of neem by anther culture, at early-to-late uninucleate stage of pollen. Callusing from anthers was induced on MS basal medium (with 9 % sucrose) supplemented with 2, 4-D (1 μ M), NAA (1 μ M), and BAP (5 µM). The calli multiplied best on MS medium (with 3 % sucrose) supplemented with 2, 4-D (1 μ M), and Kinetin (10 μ M). MS supplemented with BAP (5 μ M) was optimum for regeneration from younger calli (75 % cultures differentiated shoots), but older calli showed the best regeneration at 7.5 µM BAP. As per histological investigations, in four-week-old cultures the anther-wall cells had started dividing, while the microspores appeared unchanged. However, in eight-week-old cultures, it was observed that the entire anther locules were filled with microcalli. Calli maintained on MS medium containing 2, 4-D exhibited good regeneration potentiality, but those calli that were maintained on MS medium containing 2, 4-D (1 µM) and Kinetin (10 μ M), retained the regeneration potential for a longer period. Elongation of shoots was achieved at a lower concentration of BAP at 0.5 µM. These shoots were multiplied by forced axillary branching and were rooted through in vitro on $\frac{1}{4}$ strength of MS medium supplemented with IBA (0.5 µM). The plants were subsequently established in soil. Of the plants that regenerated from anther callus, 60 % were haploids (2n=x=12), 20 % were diploids (2n=2x=24) and 20 % were aneuploids (2n=2x-2=22). Srivastava and Chaturvedi (2011) reported a new improved method of haploid production from androgenic cultures of Azadirachta indica A. Juss. In this investigation, the best callus induction response was obtained in the induction medium with 12 % sucrose concentration on MS+2,4-D (1 μ M)+NAA $(1 \mu M)$ +BAP (5 μM). Maximum shoot regeneration frequency obtained was 98.5 % with an average of 8.5 shoot-buds/ explant on MS+BAP (2.2 µM)+NAA $(0.05 \text{ }\mu\text{M})$, as against 75 % with cultures forming an average of 4.5 shoot-buds/ explants on MS+BAP (5 μ M) in the earlier report of anther culture of neem by Chaturvedi et al. (2003). Cytological analysis of the calli and regenerants confirmed their haploid status with the chromosome number as 2n=x=12.

To date, there is only one single report on in vitro ovary culture of neem by Srivastava et al. (2009). However, the regenerated plants were diploid in nature. Unfertilized ovaries obtained from closed flower buds of an adult 54-year-old neem tree were used as explants. Maximum shoot regeneration (78 %) was observed when calli, induced from ovaries of 4 mm size flower buds and proliferating on MS+2,4-D (0.5 μ M), were transferred to MS containing BAP (5 μ M). Histological analysis revealed that 4 mm sized flower buds corresponded to a 2-nucleate stage of the embryo sac. The best medium for inducing calli from unfertilized ovaries was MS medium with 9 % sucrose, 2,4-D(1 μ M)+BAP(5 μ M). The shoots were multiplied by forced axillary branching on MS medium supplemented with BAP (1 μ M) and Casein hydrolysate (250 mg/l). The shoots were rooted on ¹/₄ strength MS medium supplemented with IBA (0.5 μ M) at a frequency of 79 %. The plants were subsequently hardened with transplantation rate of 81.8 %. Although all the regenerated plants were diploid, an efficient protocol for ovary culture of neem was established which can be used as a platform to produce gynogenic haploids of neem in future.

4.2 Camellia spp (Family: Theaceae)

Camellia sinensis (L.) O. Kuntze commonly referred to as tea, belonging to the family Theaceae, is an evergreen, perennial, cross-pollinating plant. It grows naturally as tall as 15 m; however, it is usually trimmed to below 2 m when cultivated for its leaves (Mondal and Parathi 2003). The cultivated taxa of tea consists of three main natural hybrids. They are *Camellia sinensis* (L.) O. Kuntze or China type, *Camellia* assamica (Masters) or Assam type and Camellia assamica sub spp lasiocalyx (Planchon ex Watt.) or Cambod or Southern type. Tea is the oldest non-alcoholic caffeine containing beverage in the world. The medicinal properties of tea are also widely acclaimed. It is an important socio-economic crop that plays a significant role in the foreign exchange of developing countries like India. Polyphenols account for about 25-35% of the total dry weight of freshly plucked tea leaves of which two-thirds is contributed by catechins alone (Saravanan et al. 2005). Research progress on tissue culture of tea has been rather slow for the last 20 years because tea is less amenable to these techniques, due to the high level of polyphenols, the presence of systemic bacterial contamination and its recalcitrant nature in tissue culture medium (Dood 1994).

The initial attempts to produce haploids through anther culture of tea were pioneered by Katsuo (1969) and Okano and Fuchinone (1970). They obtained roots from anther-derived callus. Later, Hoken Toi (1981) also achieved similar response on medium supplemented with NAA (9.67 μ M) and Kinetin (10.1 μ M) (Mondal et al. 2004). However, it was Chen and Liao in 1982, who produced complete plantlets from tea anthers of cultivar Fuyun No-7 out of nine different tea cultivars on which they worked. The plantlets were obtained when the anthers were cultured on N₆ medium supplemented with Kinetin (9.3 μ M), 2,4-D (2.26 μ M), L-Glutamine (800 mg/l), and Serine (100 mg/l), followed by sub culturing on N₆

medium supplemented with Zeatin (9.12 µM), adenine (148 µM) and Lactoalbumin hydrolysate (10 mg/l). On this medium, the calli continued to proliferate either into shiny mass or shoots. These shoots were subsequently rooted on medium containing IAA (0.57 μ M). While three out of four plants were haploids, the rest were an euploids with a chromosome number 2n = 18. Later, in (1992, Saha and Bhattacharya reported formation of globular structures in tea which failed to differentiate further on MS medium (with 7 % sucrose). This was supplemented with NAA (0.53 µM), 2, 4-D (0.45 µM), Kinetin (0.46 µM), and Glutamine (400 mg/l). However, the differentiation of true pollen embryos and regeneration of haploid plants were described by Raina and Iyer in (1992 and Shimokado et al. (1986). Pedroso and Pais (1994) tested 17 different media combinations based on MS and N₆ with various concentrations of carbon source, growth regulators and amino acids, such as Serine and Glutamine for C. japonica. The embryogenic calli from microspores were obtained when isolated microspores were cultured on 2, 4-D $(4.53 \,\mu\text{M})$, and Kinetin $(0.46 \,\mu\text{M})$ and subsequently on MS medium supplemented with BAP (2.22 uM). However, further growth ceased at maturation stage. Seran et al. (1999) reported the highest response in terms of micro calli formation in a Sri Lankan tea clone TRI-2043 (78-98 %) out of five different clones selected for study. On ¹/₂ MS+2, 4-D (9.06 µM)+Kinetin (4.65 µM)+IAA (5.71 µM) under dark condition, 98 % anther response was achieved. Determination of ploidy levels in the callus cells showed that the frequency of haploid cells was greater (68 %) in comparison to diploid cells (6%). However, plantlets could not be regenerated. Mishra and Chaturvedi (unpublished) has obtained for the first time microspore embryogenesis in anther cultures of TV clones of tea (TV19 and TV21) via an intervening callus phase. Within three weeks of culture, the anthers were swollen and after six weeks longitudinal dehiscence occurred along the anther wall (Fig. 8.2a). followed by emergence of transparent white callus from within (Fig. 8.2b). In both TV19 and TV21, calli were obtained when anthers were subjected to dark incubation at 25 °C. After successful induction of callus, embryogenesis in these calli and embryo germination occurred when cultures were transferred to light conditions (Fig. 8.2c, d).

So far, there have been no reports on in vitro gynogenesis in tea. Hazarika and Chaturvedi (unpublished data) have obtained for the first time, successful callus induction from unpollinated ovules of tea when thin sections of ovary cultures were subjected to various regimes of temperature and light treatments. Within a week of the culture, the ovules swelled to almost double their original size (Fig. 8.3a) and callusing was observed. In a few cultures, white, friable callus tissue emerged from within the cultured ovules (Fig. 8.3b) whereas in others profuse callusing from the entire sections occurred. The nature of callus obtained from within the ovules was friable and it was white in colour. Histological study of the ovary slice sections at the time of culture showed the presence of a mature egg cell within the embryo sac of the unpollinated ovules. Following successful callus induction, formation of dark green nodulated structures occurred on various media compositions. Histological investigations revealed the formation of tracheids in the callus tissue.

191



Fig. 8.2 a Six-week-old anther culture of tea, showing longitudinal rupturing of anther walls (220X). b 10-week-old anther culture of tea. Note the emergence of calli from inside the anther walls (90X). c Embryogenesis occurring in anther-derived callus of tea, after eight weeks of culture. Embryos at various stages of development can be seen in the Fig. (4.5X). d A germinated embryo of tea (6X)

4.3 Coffea spp (Family: Rubiaceae)

The coffee plant is a woody perennial, evergreen, dicotyledon that grows relatively high, therefore, more accurately described as a coffee tree. Coffee is a brewed drink prepared from roasted seeds, called coffee beans, of the coffee plant. While there are several different coffee species, the two main cultivated species are, *Coffea arabica*, known as Arabica coffee accounting for 75–80 % of the world's production and *Coffea canephora*, known as Robusta coffee, accounting for about 20 % of the total production. Genetic improvement of coffee, being an important commercial crop, is essential for improving the production and quality of coffee. Development of haploids via anther culture facilitates the production of homozygous plants in one generation and opens the way for new breeding strategies.



Fig. 8.3 a A slice section of tea ovary culture, showing swollen ovules, after two weeks of culture (8X). **b** Four-week-old cultures showing bursting of ovules and emergence of callus from within (8X)

Anther culture studies are quite limited in coffee. The first attempt to produce haploid coffee plants was made in *Coffea arabica* (Sharp et al. 1973). Successful reports of haploid and dihaploid plants from anther culture of coffee have been published (Montes 1981; Ascanio and Arcia 1987). Ascanio and Arcia (1987) reported the existence of a correlation between different developmental stages of anthers, the size of flower buds and the quantity of calli obtained after 90 days of culture. Further in 1994, the authors studied the effect of developmental stage and heat shock on the formation of embryogenic calli from *C. arabica* var. guernica anthers.

Unlike anther culture, isolated microspore culture allows haploid plant regeneration directly from microspores, assuring pure gametophytic origin. Colonies of haploid cells have been successfully obtained via mechanically isolated microspore cultures of C. arabica varieties Catuai and Catimor in either liquid or solid media (Carneiro 1997). Herrera et al. (2002) described, for the first time, a new approach for embryo induction and plant regeneration from C. arabica cv caturra isolated microspores, pre-treated with colchicine. Their study clearly demonstrated that colchicine could activate the androgenic response in coffee microspores. A positive androgenic development was observed only when late uninucleated or early binucleated microspores were cultured. Such response was evidenced by microspore divisions beginning after 25 days of colchicine exposure. The best androgenic response was found when microspores were pre-cultured in 100 mg/l colchicine for 48 h. Flow cytometry and morphological analyses revealed that 95 % of regenerated plants were dihaploids (2n=2x=22). However, some doubled dihaploid plants (2n=4x=44) were also obtained, suggesting that not only and rogenic induction but also chromosome doubling could be expected as a result of colchicine exposure of coffee microspores.

Lanaud in 1981 reported somatic embryogenesis in ovules from *C. canephora* and established conditions for rapid multiplication and differentiation of embryo-

genic mass to produce haploid plants. In brief, anther and ovule cultures are least developed coffee tissue culture techniques despite their substantial potential in reducing the time required to produce new varieties.

4.4 Citrus sp. (Family: Rutaceae)

Citrus species represent the largest production of fruits worldwide. They are valued for their antioxidant properties and are one of the richest sources of vitamin C and essential oils.

The first report of development of haploid seedlings of C. natsudaidai was reported by Karasawa (1971) by the application of gamma rays. Thereafter, there have been several reports on production of haploids via in vivo crossing (Oiyama and Kobayashi 1993), gynogenesis induced by in vitro pollination with pollen from a triploid plant (Germanà and Chiancone 2001) and by in vivo parthenogenesis (Germanà 2006). Anther culture technique has been used to obtain haploid calli, embryoids and plantlets with limited efforts in a few Citrus species like C. madurensis, C. limon, C. deliciosa x C. paradise (Chen et al. 1980; Germanà et al. 1991; Germanà and Reforgiato 1997). However, extensive research has been carried out on C. clementina. Since the first embryogenic calli and haploid plantlets were obtained by anther culture in C. clementing Hort. Ex Tan. cv Nules (Germanà et al. 1994), many studies have been carried out to improve the androgenic response in Citrus species by the use of different combinations of plant growth regulators (Hidaka et al. 1979; Chaturvedi and Sharma 1985; Geraci and Starrantino 1990; Germanà et al. 1994, 2000a, b, 2005). Germanà and Chiancone (2003) proposed an improved and detailed protocol for haploid induction through anther culture of C. clementina Hort. Ex Tan. cv Nules by evaluating a number of factors that affect androgenesis. They observed that temperature pre-treatment of flower buds between 4 °C and 25 °C for 14 days was more favourable to induce embryogenic callus and embryoids in anther cultures. Anthers were excised from pre-treated flower buds and cultured on N₆ medium (Chu 1978) supplemented with Nitsch and Nitsch vitamins (Nitsch and Nitsch 1969), galactose (9000 mg l^{-1}), lactose (18,000 mg l^{-1}), coconut water (5 % v/v), casein hydrolysate (500 mg l⁻¹), L-glutamine (200 mg l⁻¹), biotin $(0.5 \text{ mg } l^{-1})$, ascorbic acid (500 mg l^{-1}), NAA (0.11 μ M), 2,4-D (0.09 μ M), Kinetin (4.6 µM), BAP (2.2 µM), Zeatin (2.28 µM), TDZ (0.45 µM), and GA₂ (1.45 µM) in dark for 15 days before being shifted to diffuse light at 25 ± 2 °C. With this protocol, 1.9 % anther cultures showed embryoid development and a total of 570 and 1,000 embryoids developed in Nules and SRA 63 cultivars, respectively. Direct gametic embryogenesis without callus formation was observed in 7 % responsive anther cultures of the cv Nules and in 45 % of the responsive anther cultures of the cv SRA 63. The embryoids were later germinated on MS medium containing GA, $(2.89 \,\mu\text{M})$ and NAA $(0.05 \,\mu\text{M})$. Recently, Chiancone et al. (2006) studied the effect of polyamines, spermidine and putrescine, with an aim to further improve the rate of embryogenic callus and embryoid induction in anther cultures of C. clementiana cv Nules. The addition of 2 mM spermidine to the suggested medium of Germanà and Chiancone (2003) stimulated gametic embryogenesis in 4 % cultures whereas putrescine did not influence embryo production. Flow cytometric analysis revealed that the regenerants were mostly trihaploids; few were doubled-haploids while none of them were haploids.

4.5 Malus domestica Borkh. (Family: Rosaceae)

Malus domestica Borkh. (Family: Rosaceae) or apple is an important temperate fruit tree. It is one of the most widely cultivated tree fruits in the world. In vitro approaches to induce haploids in apple have been rather limited in comparison to other plant species (Höfer and Lespinasse 1996). Anther culture in apple was pioneered by Japanese scientists at the beginning of the 1970s. They induced calli capable of root formation. Subsequently, several working groups initiated haploid induction in apple by anther culture. Induction of embryogenesis and plant formation has been reported in apple from anther cultures (Fei and Xue 1981; Xue and Niu 1984; Höfer 1995). Although regeneration from embryos is reproducible via adventitious shoot formation, the induction of embryogenesis from cultured apple anthers is still low and highly dependent on genotype (Höfer 1995, 1997). Kolova et al. (1994) obtained multinuclear structures and Bouvier (1993) obtained callus formation from isolated microspore cultures. Höfer et al. (1999) reported, for the first time, the induction of embryogenesis and plant formation from isolated apple microspores at late uninucleate stage of development. The authors reported that several factors were responsible for successful androgenesis in apple, especially a combination of starvation and cold treatment. The effect of starvation of buds or microspores for 1-2 days at 4 °C or 27 °C was found to be very important for androgenesis in apple. Starvation is an effective stress treatment which has yielded successful embryogenic induction in tobacco (Kyo and Harada 1986; Touraev et al. 1996a) and wheat microspores (Touraev et al. 1996b). The requirement of heat or cold pre-treatment has been shown to vary from plant to plant (Höfer et al. 1999). The induction medium and genotype of the donor plant strongly affect embryogenic capacity of isolated microspores. "Alkmene" and "Rene" are the cultivars of apple in which the highest embryo induction via anther culture was achieved. Success with "Rene" has been reflected in many experiments related to androgenesis in apple. They attained 17 % embryo induction from microspore cultures as compared to only 7 % induction from anther cultures on modified N₆ basal medium devised by Chu et al. (1990) in wheat. The embryos were germinated on MS+TDZ (0.45 μ M). Höfer (2004) reported that an increase in the frequency of embryo induction is possible up to 10 times by microspore culture depending on the genotype. Starvation treatment, induction medium, maltose concentration, type of culture vessel, microspore density and genotype influenced embryo induction in apple Höfer (2004).

4.6 Morus sp. (Family: Moraceae; Common name: Mulberry)

Mulberry is a vital crop for sericulture industry and is native to warm temperate subtropical regions of Asia, Africa, Europe, and the Americas, with the majority of the species native to Asia alone. Mulberry leaves are ecologically important as the only food source for the silkworms (*Bombyx mori*), the pupa/cocoon which are used to make silk. Apart from its application as a food source for the silkworms, the medicinal properties of mulberry sp. are also widely acclaimed. Like many other woody genera, mulberry is highly heterozygous and, therefore, production of haploids and doubled haploids through anther culture is highly beneficial.

Several authors have reported androgenesis in mulberry which dealt with studies performed on induction of division in pollen culture (Katagiri 1989), effects of sugars and alcohols on pollen division (Katagiri and Modala 1991), embryo differentiation (Sethi et al. 1992) and production of haploid plantlets from anther culture (Shoukang et al. 1987). However, detailed studies were conducted by Jain et al. (1996) who evaluated the effect of temperature and Kinetin pre-treatment on induction of androgenic callus in anther cultures of mulberry. It was revealed that cold pre-treatment given to flower buds at 4 °C for 24 h increased the percentage of callus originating from anther cultures. The anthers split and produced embryogenic callus on Modified Bourgin (MB) medium (Qian et al. 1982) with 8 % sucrose and supplemented with NAA (2.68 µM) and BAP (4.44 µM). Upon transfer of the calli to MB basal medium supplemented with NAA (2.68 µM), BAP (2.22 µM), 2,4-D (4.12 µM) and Polyvinyl pyrrolidone (PVP) (1.0 mg/l), embryos were induced which later developed roots upon removal of 2, 4-D from the medium. Finally, the embryoids germinated precociously without developing cotyledons, but forming elongated shoots. Rhizogenesis was induced when calli were subcultured on MB medium containing NAA and BAP (each 0.5 mg/l with reduced myoinositol (75 mg/l). The cytological study of the induced roots from the calli revealed the haploid nature of the callus tissue.

To raise haploids from female clones of mulberry, ovary culture is the only possible approach because inbreeding and anther culture are not applicable (Thomas et al. 1999). In *ab initio* individual ovary cultures of *Morus alba*, the growth was very poor and gynogenic plants were never formed, suggesting that unfertilized ovaries required some stimulus from the parental tissue for the initial growth. This is in contrast to the observations of Lakshmi Sita and Ravindran in 1991 who observed gynogenesis in *ab initio* ovary cultures of *Morus indica*. These authors cultured individual ovaries before or after fusion to form sorosis under field conditions without taking any measure to prevent chance pollination. The inflorescence of mulberry is a catkin in which ovaries are loosely arranged. After fertilization, the ovaries enlarge and fuse to form a solid looking fruit called 'sorosis'. It was possible that some of the gynogenic plants were haploid, others were diploid. (Thomas et al. (1999) followed a two-step protocol for ovary culture of mulberry. They raised nodal segment cultures on MS+BAP (5 μ M) which developed axil-

lary shoots and one or more inflorescence from the axil or leaves in three weeks. Inflorescence segments, each bearing 4–5 florets, from four-week-old cultures of nodal segments, were planted on MS+BAP (8.5 μ M)+2,4-D (4.5 μ M) and after three weeks individual ovaries were transferred to MS+2,4-D (4.5 μ M)+Glycine (0.5 mg/l)+Proline (0.2 mg/l). In this treatment, 16 per cent of the ovaries developed a gynogenic seedling. After the gynogenic plants are transplanted in soil, out of 20 plants examined, 12 showed haploid number of chromosome (2n=x=14) and the other eight were aneuploids with 13–17 chromosomes in their root tip cells.

4.7 Populus sp. (Family: Salicaceae)

Poplars form a very important part of basic forest biology and are economically important trees cultivated for their high wood quality that finds use in paper industry and energy production. The genus Populus consist of more than 30species, occurring throughout the forests of temperate and cold regions of northern hemisphere. The species *P. ciliata*, the only native poplar of India, is endemic to the Himalayan belt and has been an important tree for the forest breeders in India. In 1950, a large number of exotic clones of *Populus* were introduced and grown in North India, mainly, for increasing the wood availability for match and plywood industries. The study of poplars is essential to complement the ever increasing knowledge database of this important tree species.

First successful report on anther culture in *Populus* is by Wang et al. (1975), who observed callus proliferation from pollen and subsequent formation of haploid plants via organogenesis from the pollen-derived calli. Later, several reports were published on production of pollen plantlets in anther cultures of poplar (Zhu et al. 1980; Ho and Raj 1985; Kim et al. 1986; Mofidabadi et al. 1995). Induction of haploids via embryogenesis occurring from cultured anthers of P. Maximowiczii was obtained by Stoehr and Zsuffa (1990). The scientists applied cold pre-treatment to the flower buds (at 4 °C for four days) prior to culture the anthers at uninucleate stage of microspores, on MS medium containing 2,4-D (2.26 µM) and Kinetin (0.46 µM). Globular calli were developed from anthers after 4-8 weeks of dark incubation in the medium at 20 °C. When the anthers with globular calli were cultured on MS medium supplemented with NAA (0.54 µM) and BAP (4.4 µM), microspore division and embryoidal structures resembling globular-to-heart-shaped embryoids were obtained. However, the embryoids germinated precociously without developing cotyledons. After transfer to MS medium with BAP (4.4 µM), adventitious shoots developed mainly from the roots. Shoots were rooted on half strength MS medium supplemented with NAA (0.13 µM). Out of 34 plants analyzed cytologically, 22 showed haploid chromosome number (n=19), one was an euploid and the rest were diploids (Srivastava and Chaturvedi 2008). Kiss et al. (2001) reported further improvement in the rate of haploid plant regeneration by increasing the rate of induction from the anthers and with sustained shoot regeneration frequency in five different genotypes of two poplar species, viz., *P. nigra* L. and *P. deltoides* Bartr. They reported that frequency of callus induction, shoot regeneration and number of shoots developed per calli are highly dependent on the genotype. The anthers were taken from floral buds which were subjected to cold pre-treatment at 4 °C for 8–14 days. For callus induction and shoot regeneration, they used the media suggested by Stoehr and Zsuffa (1990) with slight modifications. The anthers were incubated in dark for 6–7 weeks at 25 °C. Calli for shoot organogenesis were transferred to both MS or WPM medium (Lloyd and McCown 1980) supplemented with BAP (4.4 μ M), NAA (0.54 μ M), and 2.5 % sucrose and incubated in light at 23 °C. Shoots were rooted on hormone-free WPM. Of the 208 rooted plants, eight haploids, 179 diploids, four tetraploids and 17 aneuploid plants were obtained. Deutsch et al. (2004) reported haploid plant regeneration via embryogenesis from isolated immature pollen culture of two poplar hybrids (*Populus nigra* L.× hybrid 'Aue1' and 'Aue2') and employed microsatellite marker analysis to confirm the ploidy level of the regenerants.

5 Analysis of Ploidy Level

Analysis of Ploidy level is an integral part of haploid production programme which can be carried out efficiently either by chromosome counting during mitotic and meiotic cell division or by flow cytometry.

5.1 Chromosome Counting

Counting of mitotic chromosomes is easier and faster. Root tips are the most convenient source of mitotic cells. When roots are not available, young axillary buds, leaves or callus can be used. The cytological procedures of chromosome preparation and staining are different and need to be modified depending on the plant species. The cytological procedures for visualizing chromosomes in woody plants may not be the same as in herbaceous species. Nevertheless, basic principles for handling the mitotic chromosomes of all plant species are similar and consist of collection of material, fixation and chromosome staining. The most critical step for chromosome counting involves proper chromosome squash preparation. It is very important to obtain sufficient well spread metaphase plates and proper physical separation of the chromosomes. However, method of chromosome staining applied for ploidy level analysis depends on plant species and chromosome size (Maluszynski et al. 2003a).

Protocol for cytological investigation in neem (Chaturvedi et al. 2003) Fixation:

• Healthy root-tips (1 cm) from plantlets were collected at 10.30 am and rinsed with tap water.

- The root-tips were pre-treated with 0.02 % 8-Hydroxyquinoline at 4 °C for 4 h.
- Following which, the root-tips were fixed in a modified Carnoy's fluid containing absolute alcohol-chloroform-Glacial acetic acid-Methanol (7:3:1:1) for 48 h.
- Finally, the root-tips were preserved in 70 % ethanol at 4 °C.

5.2 Chromosome Staining

- The root-tips from fixed roots were excised and placed in a mixture of nine drops of 2 % aceto-orcein and one drop of 1N HCl in a watch glass and warmed gently.
- After cooling, the individual root-tips were placed in a drop of aceto-orcein on a glass slide, covered with a cover slip, warmed gently and squashed.

5.3 Flow Cytometric Analysis

Flow cytometry using DNA selective flourochromes has been considered to be a fast and reliable method for the measurement of nuclear DNA content in recent years (Muirhead et al. 1985; Thorthwaite et al. 1985; Dolezel et al. 2007). Unfortunately, its application in plant biology has been overdue, largely owing to the fact that flow cytometry requires single cell suspension (Shapiro 1985). As plant cells usually exist in complex tissues, methods had to be developed for the preparation of such suspensions. Although flow cytometry is an extremely efficient technique with high degree of accuracy, the preparation of high quality plant samples for ploidy analysis remains a vital issue. For determination and interpretation of the haploid status of the regenerants, firstly tissue of known ploidy is analyzed followed by the unknown tissue whose ploidy is to be analyzed.

5.4 Protocol for Flow-Cytometric Analysis in Tea

5.4.1 Preparation of Nuclear Sample

- Well developed calli obtained after two months of culture initiation were used for ploidy analysis.
- Extraction of nuclei and the analysis were carried out via fine chopping of the hard calli placed in 1 ml ice-cold woody plant buffer. The woody plant buffer was prepared by mixing 0.2 M Tris HCl, 4mM MgCl₂.6H₂O, 2mM EDTA Na₂.2H₂O, 86mM NaCl, 10 mM Sodium Metabisulfite, 1 % Triton X-100 (v/v), and 2 % PVP-10 (w/v) according to the protocol of Loureiro et al. 2007 with

slight modifications. The pH of the buffer was adjusted to 7.5, filtered through 0.22 μ m Polyvinylidene fluoride (PVDF) membrane filter and stored at 4 °C.

- The suspension containing the nuclei was mixed gently by pipetting up and down, softly, several times, followed by filtering of the homogenate through a $30 \ \mu m$ nylon mesh.
- The nuclear suspension was stained with Propidium iodide at a concentration of 50 μg/ml.
- Simultaneously RNAse at a concentration of 50 μ g/ml was also added to the nuclei.

Ploidy analysis:

- The ploidy level was determined using a FACs Calibur cytometer (Becton-Dickinson, USA).
- All measurements were carried out in triplicate using fresh tea leaves as an external standard. Using instrument gain (photomultiplier voltage and amplitude gain), the position of peak G₁ nuclei of the reference sample was established on channel 200 on a 1,024 scale following which the instrument settings were kept constant and the unknown samples were run under the same parameters. The mean channel number of the unknown sample G₁ peak was determined and the DNA Ploidy was calculated according to the relationship:

Sample Ploidy (integer) = Reference Ploidy X (mean position of the G_1 sample peak/mean position of the G_1 reference peak)

6 Diploidization of Haploids

Doubled haploids are of immense importance for genetic studies and in crop breeding programmes. Selected doubled haploid lines are used for production of commercial hybrids (Chase 1974). To obtain fertile, homozygous diploids for analyzing the progenies and the breeding behaviour of the pollen plants, the chromosome complement of the haploids must be duplicated. This is because haploids may grow normally up to the flowering stage, but in the absence of homologous chromosomes, meiosis is abnormal and consequently, viable gametes will not be formed and, hence, they are sterile. Spontaneous duplication of chromosomes in pollenderived plants has been observed, but its frequency is very low.

In vegetable crops, doubled haploids are used prominently as parents for F_1 hybrid seed production. Similarly, in medicinal and aromatic plants, doubled haploids for F_1 production have the potential to make significant advances in providing high, stable and predictive yields of raw biochemicals to be processed by pharmaceutical and neutraceutical industries (Ferrie 2007). In case of cross pollinating species and hybrids, pure homozygous lines are highly desirable. Although conventional breeding method to produce homozygous diploids is well

established, it requires 7–8 repeated inbreeding cycles which in turn is time-consuming and labour intensive. Moreover, this approach is difficult in self-incompatible male sterile plants and tree species with long gestation period. Also, the pure lines obtained after several generations of self-pollination may not be 100 % homozygous (Germanà 2006). Production of fertile doubled haploid lines in S. cereale (Immonen and Anttila 1996) and Festuca and Lolium (Nitzche 1970) are of immense importance as these species suffer from inbreeding depression. One of the most efficient techniques for production of homozygous diploid plants in a single generation is through diploidization of the haploids. Doubled haploids have a key feature in establishing chromosome maps in a number of species, notably barley Hordeum vulgare, Orvza sativa, Brassica napus, Triticum aestivum (Forster and Thomas 2005), apart from providing a vast majority of mapped genetic markers. Molecular marker maps provide a platform for trait mapping, which is of particular interest to plant breeders. Doubled haploidy is also useful in rapid isolation and purification of selected mutants in subsequent generations. It is possible to create a population of homozygous mutant lines directly by targeting the mutation treatment at single gametic cells and then inducing embryogenesis, followed by subsequent chromosome doubling for homozygous diploid plant production (Szarejko and Forster 2006).

Doubled haploidy can be significantly enhanced by artificial means using chemicals such as colchicine, pronamide, trifluralin, oryzalin and amiprophos methyl (APM) (Wan et al. 1991). Colchicine treatment is one of the most preferred techniques for chromosome doubling, which in turn is one of the most critical steps in the doubled haploid breeding process. For Nicotiana tabacum, a 0.4 % solution of colchicine is recommended to diploidize the pollen plants. In practice, the young pollen derived plantlets are immersed in a filter sterilized solution of colchicine for about 96 h and then transferred to a culture medium to allow their further growth. Alternatively, the treatment is given in the form of a lanolin paste (Bhojwani and Razdan 1996). It is applied to the axils of the upper leaves and the main axis is decapitated to stimulate the axillary buds to grow into diploid and fertile branches. Besides bringing about chromosome duplication, colchicine treatment may also result in chromosome and gene instabilities (Burk 1970). Therefore, the frequent occurrence of spontaneous duplication of chromosomes in differentiated plant cells like cortex and pith and callus cells in long-term cultures has been exploited to raise homozygous, fertile diploids from haploid plants. In Apiaceae, for diploidization the plantlets grown in Petri dishes were taken out and the agar was removed from the roots. Subsequently, the roots were submerged in a 0.34 % (w/v) solution of colchicine for $1\frac{1}{2}$ h. The roots were rinsed in water and the plantlets were transferred to a soil-less mix and grown in the greenhouse (Ferrie et al. 2005). For Brassica species, the roots of 25-30 pollen-derived plantswere immersed in a bunch in 0.25 % (w/v) colchicine solution for 5 h in light, for diploidization. After rinsing the treated roots with distilled water, the plants were transferred to a potting mix for hardening and further growth (Bhojwani and Dantu 2010).

7 Application of Haploid Production

7.1 Development of Pure Homozygous Lines

Homozygous true breeding cultivars are highly important for screening of high yielding lines and to produce hybrid vigour as a method of crop improvement. Obtaining homozygous diploid plants by conventional methods is difficult in perennials. From several decades to over a hundred years are required to obtain a pure line by means of successive inbreeding throughout many generations. The seed set by inbreeding in many trees is very low, usually only a few of ten thousandth or sometimes no seed can be obtained at all; therefore, it is impractical to obtain pure lines by inbreeding (Chen 1986). Moreover, conventional method of haploid production by inbreeding is impossible if the plant is strictly cross-pollinating in nature. On the other hand, homozygous diploid plants can be achieved in a single generation by diploidization of in vitro raised haploids through colchicine treatment.

7.2 Genetic Studies

Because of the lack of accurate material in research work, the progress in the study of genetics in trees is much slower than that in annual herbaceous plants. The genetics of a lot of important traits in economically important plant species has not been clearly demonstrated as yet. As a result, it is still unknown whether the desirable characters of the parents will appear in their progenies. Only when crossing between different homozygous diploid plants is carried out, we can gain a clear idea of dominance of genes controlling various characteristics and that these characteristics are either monogenic or polygenic (Chen 1986). Furthermore, if we can use the haploid plants as samples of gametes, then we can obtain directly the recombination value between genes. Moreover, we can also use the haploid plants to study chromosome homology within genome or between genomes.

7.3 Gametoclonal Variation

The "gametoclonal variation" arises among plants regenerated from cultured gametic cells consisting of differences in morphological and biochemical characteristics as well as in chromosome number and structures that are observed. Besides yielding haploids, in vitro androgenesis helps in the screening of gametophytic variation at plant level. Pollen grains within an anther form a highly heterogeneous population because they are the product of meiosis which involves recombination and segregation. Therefore, each pollen plant is genetically different from the other. The gametoclonal variants being hemizygous in nature, express also the recessive characteristics in the R_0 plants (Bhojwani and Razdan 1996). Different sources of variation can explain gametoclonal variation such as new genetic variation induced by cell culture procedures, from segregation and independant assortment, chromosome doubling procedures, etc., (Morrison and Evans 1987; Huang 1996).

7.4 Induction of Mutations

In general, a majority of induced mutations are recessive and, therefore, are not expressed in diploid cells due to the presence of dominant allele. Since haploid plants have only one set of chromosomes, their dominant and recessive characteristics can be seen simultaneously on separate plants. It is extremely advantageous to provide a convenient system for the induction of mutations and selection of mutants with desirable traits in the absence of their dominant counterparts (Bhojwani and Razdan 1996).

7.5 Obtaining New Genotypes with Alien Chromosomes

The technique of interspecific and intergeneric hybridization can be combined with anther culture techniques (Thomas et al. 2003) for obtaining new genotypes with alien chromosomes. Thus new genotypes with various reconstructed chromosome complements can be obtained after their successful chromosome doubling.

7.6 Genetic Manipulation

As microspore culture is a single cell system, it makes selection at the single cell level possible and, furthermore, offers new prospects for genetic manipulation like mutagenesis and transformation. Direct gene transfer by microinjection offers the possibility of transgenic plant formation by using isolated pollen culture having high regeneration efficiency (Kasha and Maluszynski 2003). Moreover, if transgenes can be incorporated into the haploid microspore genome prior to DNA synthesis and chromosome doubling, the doubled haploids may also be homozygous for the transgenes. Thus isolated microspores not only provide a good target for bombardment, but also are readily amenable to transgene in vitro selection. Jahne et al. (1994) were the first to achieve plants homozygous for the transgenes using biolistic bombardment of barley microspores.

7.7 Genomics

Doubled haploids play a vital role in genomics, especially, in the integration of genetic and physical maps, thereby, providing precision in targeting candidate genes (Kunzel et al. 2000; Wang et al. 2001). Doubled haploids, combined with markerassisted selection, provide a short-cut in backcross conversion, a plant breeding method for improving an elite line defective in a particular trait (Toojinda et al. 1998). Expressed sequence tags may help in identification of genes that determine any trait of interest.

8 Conclusion

The haploid-derived true breeding lines provide a rapid means of achieving homozygosity, thereby speeding up the usually cumbersome and protracted conventional breeding methods for crop improvement. Today, haploids and doubled haploids have been reported in over 200 plant species (Forster et al. 2007), however, less than 10 % of these reports encompass tree species. Thus much work needs to be done in the field of in vitro haploid production. This may be feasible only with the development of novel genotype-independent methods through the study and improvement of existing protocols and by obtaining a better understanding of the haploid induction process, especially of the two main developmental switches: the induction of the male/female gametophyte to undergo division, and subsequent occurrence of embryogenesis. The recent spurt in the development of ultra high throughput technologies of advanced genomic, transcriptomic, proteomic and imaging tools hold great promise for identification and analysis of genes that might be playing important roles in the haploid induction process. This will immensely help in the understanding of these processes and towards development of highly efficient protocols for production of haploids, resulting in the overall improvement of highly desirable and recalcitrant plant species. Through this chapter, we intend to provide a clear and simple overview of haploidy with the purpose to stimulate interest among scientists working in the related area and the potential application of the various in vitro techniques used in haploid production intended for crop improvement.

References

- Achar PN (2002) A study of factors affecting embryo yields from anther culture of cabbage. Plant Cell Tiss Org 69:183–188
- Agarwal PK, Agarwal P, Custers JBM, Liu C, Bhojwani SS (2006) PCIB an antiauxin enhances microspore embryogenesis in microspore culture of Brassica juncea. Plant Cell Tiss Org 86:201–210
- Alan AR, Martha AM, Aigars B, Cobb Ed, Elizabeth D (2003) Early production of gynogenic plants from hybrids of *Allium cepa* L. and A. roylei Stearn. Plant Sci 165:1201–1211
- Alan AR, Brants A, Cobb E, Goldschnied PA (2004) Fecund gynogenic lines from onion (*Allium cepa* L.) breeding materials. Plant Sci 167:1055–1066
- Andersen SB (2005) Haploids in the improvement of woody species. In: Palmer CE, Keller WA, Kasha K (eds) Haploids in crop improvement II, Springer, Heidelberg, 56:243–257
- Armstrong TA, Metz SG, Mascia PN (1987) Two regeneration system for the production of haploid plants from wheat anther culture. Plant Sci 51:231–237

- Arnison PG, Donaldson P, Jackson A, Semple C, Keller W (1990) Genotype-specific response of cultured broccoli (Brassica oleracea var. italica) anthers to cytokinins. Plant Cell Tiss Org 20:217–228
- Ascanio ECE, Arcia MMA (1987) Haploids from anther culture in *Coffea arabica* L. In: Abstracts of the international congress of plant tissue culture-tropical species, Bogotá, Colombia, p 68
- Ashok Kumar, HG, Ravishankar BV, Murthy HN (2004) The influence of polyamines on androgenesis of *Cucumis sativus* L. Eur J Hortic Sci 69:201–205
- Bagni N, Tassoni A (2001) Biosynthesis, oxidation and conjugation of aliphatic polyamines in higher plants. Amino Acids 20:301–317
- Ball S, Zhou HP, Konzak CF (1993) Influence of 2, 4-D, IAA, and duration of callus induction in anther cultures of spring wheat. Plant Sci 90:195–200
- Bajaj YPS (1990) In vitro production of haploids and their use in cell genetics and plant breeding. In: Bajaj YPS (ed) Biotechnology in agriculture and forestry, part I. Haploids in crop improvement, Springer, Berlin, 12:1–44
- Barclay IR (1975) High frequency of haploid production in wheat (*Triticum aestivum*) by chromosome elimination. Nature 256:410–411
- Bhat JG, Murthy HN (2007) Factors affecting in vitro-gynogenic haploid production in niger (*Guizotia abyssinica* (L.f.) Cass.). Plant Growth Regul 52:241–248
- Bhojwani SS, Dantu PK (2010) Haploid plants. In: Plant cell culture edited by Michael R. Davey and Paul Anthony, A John Wiley & Sons, Ltd., Publication
- Bhojwani SS, Razdan MK (1996) Plant Tissue Culture: Theory and practice, a revised edition. Elsevier. Studies in Plant Sci 5:167–214
- Blakeslee AF et al (1922) A haploid mutant in the Jimson weed, Datura stramonium. Science. 55:646–647
- Bouvier L, Zhang YX, Lespinasse Y (1993) Two methods of haploidization in pear, Pyrus communis L., greenhouse seedling selection, in situ parthenogenesis induced by irradiated pollen. Theor Appl Genet 87:229–232
- Burk LG (1970) Green and light-yellow haploid seedlings from anther of sulphur tobacco. J Hered 61:279
- Cagnet-Sitbon M (1980) Recherches peliminaires sur la production d'haploides de Gerbera jamsonii par cultured'antheres el d'ovules non fecondes in vitro. Th6se de 3~cycle, Am~lior. Plantes, Universit6 Paris-Sud, Centred'Orsay
- Cai DT, Chen DT, Zhu H, Jin Y (1988) In vitro production of haploid plantlets from the unfertilized ovaries and anther of Hubei photosynthetic genic male sterile rice (HPGMR). Acta Bio Exp Sin 21:401–407
- Campion B, Azzimonti MT, Vicini E, Schiavi M, Falavigna A (1992) Advances in haploid plants induction in onion (*Allium cepa* L.) through *in vitro* gynogenesis. Plant Sci 86:97–104
- Carneiro MF (1997) Determination of ploidy level of coffee plants regenerated through anthers culture by using the chloroplast counting technique In: Abstracts of the 17th Colloquium of Coffee Science Association (ASIC), Nairobi, Kenya, 20–25 July 1997, pp 105
- Cardy BJ (1986) Production of anther-derived doubled haploids for breeding oilseed rape (*Brassica napus* L.). PhD Thesis, University of Guelph
- Chalak L, Legave JM (1997) Effects of pollination by irradiated pollen in Hayward kiwifruit and spontaneous doubling of induced parthenogenetic trihaploids. Sci Hortic Amsterdam. 68:83–93
- Chand S, Sahrawat AK (2007) Embryogenesis and plant regeneration from unpollinated ovary culture of Psoralea corylifolia. Biol Plant 51:223–228
- Chase SS (1974) Utilization of haploids in plant breeding: Breeding diploid species. In: Kasha KJ (ed) Haploids in Higher Plants, Guelph: University of Guelph Press, pp 1974
- Chaturvedi HC, Sharma AK (1985) Androgenesis in Citrus aurantifolia (Christm.) Swingle. Planta 165:142–4
- Chaturvedi R, Razdan MK, Bhojwani SS (2003) Production of haploids of neem (Azadirachta indica A. Juss.) by anther culture. Plant Cell Rep 21:531–7
- Chen Z (1986) Induction of androgenesis in woody plants. In: Han H, Hongyuan (eds) Aploids in higher plants in vitro. China Academic Publishers, Springer-Verlag, pp. 42–66

- Chen Z, Liao H(1982) Obtaining plantlet through anther culture of tea plants. Zhongguo Chaye 4:6–7
- Chen Z, Wang H, Liao H (1980) The induction of Citrus pollen plants in artificial media. Acta Genet Sin 7:189–92
- Chen Z, Qian C, Qin M, Xu X, Xiao Y (1982) Recent advances in anther culture of Hevea brasiliensis (Muell.-Arg.). Theor Appl Genet 62:103–8
- Chen JF, Cui L, Malik AA, Mbira KG (2011) In vitro haploid and dihaploid production via unfertilized ovule culture. Plant Cell Tiss Org 104:311–319
- Chiancone B, Tassoni A, Bagni N, Germanà MA (2006) Effect of polyamines on in vitro anther culture of Citrus clementina Hort. ex Tan. Plant Cell Tiss Org 87:145–53
- Choo TM, Reinbergs E, Kasha KJ (1985) Use of haploids in breeding barley. Plant Breed Rev 3:219–252
- Chu C (1978) The N6, medium and its applications to anther culture of cereal crops. Proc. Symp. Plant Tissue Cult. Peking. Science Press, pp 43–50
- Chu CC, Hill RD, Brule-Babel AL (1990) High frequency of pollen embryoid formation and plant regeneration in *Triticum aestivum* L. on monosaccharide containing media. Plant Sci 66:255–62
- Collins GB, Grosser JW (1984) Culture of embryos. In: Vasil IK (ed) Cell culture and somatic cell genetics of plants, Vol. 1: Laboratory procedures and their applications. Academic Press, New York, pp 241–257
- Custódio L, Carneiro MF, Romano A (2005) Microsporogenesis and anther culture in carob tree (*Ceratonia siliqua* L.). Scientia Horticulturae 104(1):65–77
- Datta SK (2005) Androgenic haploids: Factors controlling development and its application in crop improvement. Current Sci 89(11):1870–1878
- Deutsch F, Kumlehn J, Ziegenhagen B, Fladung M (2004) Stable haploid poplar callus lines from immature pollen culture. Physiol Plant 120(4):613–22
- Dolezel J, Greilhuber J, Suda J (2007) Estimation of nuclear DNA content in plants using flow cytometry. Nat protoc 2(9):2233–2244
- Dood AW (1994) Tissue culture of tea (*Camellia sinensis* (L.) O.Kuntze) A review. Inter J Trop Argie 12(3 & 4):212–247
- Dunwell JM (1986) Pollen, ovule and embryo culture, as tools in plant breeding. In: Withers LA, Alderson PG (eds) Plant tissue culture and its agricultural applications. Butterworths, London, pp 375–404
- Dunwell JM (2010) Haploids in flowering plants: Origins and exploitation. Plant Biotechnol J 8:377–424
- Dunwell JM, Thurling N (1985) Role of sucrose in microspore embryo production in *Brassica* napus ssp. oleifera. J Exp Bot 36:1478–1491
- Fei KW, Xue GR (1981) Induction of haploid plantlets by anther culture in vitro in apple cv. 'Delicious'. Sci Agric Sin 4:41–4
- Ferrie AMR (2007) Doubled haploid production in nutraceutical species: A review. Euphytica. 158:347–357
- Ferrie AMR, Bethune T, Kernan Z (2005) An overview of preliminary studies on the development of doubled haploid protocols for nutraceutical species. Acta Physiol Plant 27:523–529
- Forster BP, Thomas WTB (2005) Doubled haploids in genetics and plant breeding. Plant Breed Rev 25:57–88
- Forster BP, Heberle-Bors E, Kasha KJ Touraev A (2007) The resurgence of haploids in higher plants. Trends in plant science 12(8):368–375
- Gamborg OL, Miller RA, Ojima K (1968) Nutrient requirements of suspension cultures of soybean root cells. Exp Cell Res 50:151–8
- Gautam VK, Nanda K, Gupta SC (1993) Development of shoots and roots in anther derived callus of Azadirachta indica A. Juss. A medicinal tree. Plant Cell Tiss Org 34:13–18
- Gelebart P, San Noeum LH (1987) Obtention de plantes haploids par culture in vitro d'ovaries et d'ovules non fecondes de Tournesol (*Helianthus annnus* L.). Agronomie 7:81–86

- Geraci G, Starrantino A (1990) Attempts to regenerate haploid plants from in vitro cultures of Citrus anthers. Acta Hortic 280:315–20
- Germanà MA (1997) Haploidy in Citrus. In: Jain SM, Sopory SK, Veilleux RE (eds) In vitro haploid production in higher plants, Kluwer, Dordrecht, 5:195–217
- Germanà MA (2006) Doubled haploid production in fruit crops. Plant Cell Tiss Org 86:131-46
- Germanà MA (2007) Haploidy. In: Khan I (ed) Citrus. Genetics, breeding and biotechnology. CABI, Wallingford, pp 167–196
- Germanà MA (2009) Haploid and doubled haploids in fruit trees. In: Touraev A, Forster B, Jain M (eds) Advances in haploid production in higher plants. Springer, Heidelberg, pp 241–263
- Germanà MA (2011) Anther culture for haploid and doubled haploid production. Plant Cell Tiss Org 104:283–300
- Germanà MA, Chiancone B (2001) Gynogenetic haploids of Citrus after in vitro pollination with triploid pollen grains. Plant Cell Tiss Org 66:59–66
- Germanà MA, Chiancone B (2003) Improvement of the anther culture protocol in Citrus clementina Hort. ex Tan. Plant Cell Rep 22:181–7
- Germanà MA, Reforgiato G (1997) Haploid embryos regeneration from anther culture of 'Mapo' tangelo (Citrus deliciosa×C. paradisi). Advan Hort Sci 11:147–52
- Germanà MA, Crescimanno FG, De Pasquale F, Wang YY (1991) Androgenesis in 5 cultivars of *Citrus limon L. Burm. f. Acta Hort 300:315–324*
- Germanà MA, Wang YY, Barbagallo MG, Iannolino G, Crescimanno FG (1994) Recovery of haploid and diploid plantlets from anther culture of Citrus clementina Hort. ex Tan. and Citrus reticulata Blanco. J Hort Sci 69(3):473–80
- Germanà MA, Crescimanno FG, Motisi A (2000a) Factors affecting androgenesis in Citrus clementina Hort. ex Tan. Advan Hortic Sci 14:43–51
- Germanà MA, Crescimanno FG, Reforgiato G, Russo MP (2000b) In: Goren R, Goldschmidt EE, (eds) Preliminary characterization of several doubled haploids of Citrus clementina cv. Nules, 535. Proc. 1st Int Symp Citrus Biotechnol. Acta Hortic pp 183–90
- Germanà MA, Chiancone B, Lain O, Testolin R (2005) Anther culture in Citrus clementina: A way to regenerate trihaploids. Austr J Agric Res 56:839–45
- Guha S, Maheshwari SC (1964) In vitro production of embryos from anthers of Datura. Nature 204:497
- Guha S, Maheshwari SC (1966) Cell division and differentiation of embryos in the pollen grains of Datura in vitro. Nature 212:97–98
- Herrera JC, Combes MC, Anthony F, Charrier A, Lashermes P (2002) Introgression into the allotetraploid coffee (*Coffea arabica* L.): segregation and recombination of the C. *canephora* genome in the tetraploid interspecific hybrid (C. *arabicax* C. *canephora*). Theor Appl Genet 104:661–668
- Hidaka T, Yamada Y, Shichijo T (1979) In vitro differentiation of haploid plants by anther culture in *Poncirus trifoliate* (L.) Raf. Jpn J Breed 29:248–254
- Ho RH, Raj Y (1985) Haploid plant production through anther culture in poplars. For Ecol Manage 3:133–42
- Höfer M (1995) In-vitro-androgenese bei Apfel. Gartenbauwissenschaft. 60:12-5
- Höfer M (1997) In vitro androgenesis in apple: Optimization of the anther culture. Acta Horticult 447:341–3
- Höfer M (2004) In vitro androgenesis in apple improvement of the induction phase. Plant Cell Rep 22:365–70
- Höfer M, Lespinasse Y (1996) Haploidy in apple. In: Jain SM, et al (eds) In vitro Haploid production in higher plants. Kluwer Academic Publishers, Netherlands, pp 259–74
- Höfer M, Touraev A, Heberle-Bors E (1999) Induction of embryogenesis from isolated apple microspores. Plant Cell Rep 18:1012–1017
- Hu H, Yang HY (eds) (1986) Haploids in higher plants in vitro. China Academic Publishers/ Springer, Beijing
- Huang B (1996) Gametoclonal variation in crop improvement. In: Jain SM, Sopory SK, Veilleux RE (eds) In vitro haploid production in higher plants, Kluwer, Dordrecht, 2:73–91

- Huang QF, Yang HY, Zhou C (1982) Embryological observations on ovary culture of unpollinated young flowers in *Hordeum vulgare* L. Acta Bot Sin 24:295–300
- Huang B, Bird S, Kemble R, Simmonds D, Keller WA, Miki B (1990) Effects of culture density, conditioned medium and feeder cultures on microspore embryogenesis in *Brassica napus*. Plant Cell Rep 8:594–597
- Immonen S, Anttila H (1996) Success in rye anther culture. Votr Pflanzenzuchtg. 35:237–244
- Inagaki NK (1990) Wheat haploid through bulbosum technique. In Bajaj YPS (ed) Biotechnology in agriculture and forestry, vol 13. Wheat. Springer, Berlin Heidelberg New York Tokyo
- Jahne A, Becker D, Brettschneider R, Lorz H (1994) Regeneration of transgenic, microsporederived Fertile barley. Theor Appl Genet 89:525–533
- Jain AK, Sarkar A, Datta RK (1996) Induction of haploid callus and embryogenesis in in vitro cultured anthers of mulberry (Morus indica). Plant Cell Tiss Org 44:143–147
- Johansson LB (1986) Effect of activated charcoal, cold treatment and elevated CO2 concentration on embryogenesis in anther culture. In: Horn W, Jensen CJ, Odenbach Schieder O (eds) Genetic Manipulation in Plant breeding. Proceedings of the International Symposium organised by Eucarpia. September 8–12, 1985, Walter de Gruyter & Co., Berlin, pp. 257–264
- Kameya T, Hinata K (1970) Induction of haploid plants from pollen grains of Brassica. Jpn J Breed 20:82–87
- Karasawa K (1971) On the occurrence of haploid seedlings in Citrus natsudaidai Hayata. Bull Sakushingakuin Junior ColWoman. 1:1–2
- Kasha KJ (ed) (1974) Haploids in higher plants: Advances and potential. The office of continuing education, University of Guelph Press, Guelph
- Kasha KJ (2005) Chromosome doubling and recovery of doubled haploid plants. In: Palmer CE, Keller WA, Kasha KJ (eds) Haploids in crop improvement II, Springer, Heidelberg, 56:123–152
- Kasha KJ, Kao KN (1970) High frequency haploid production in barley (*Hordeum vulgare* L.). Nature 225:874–876
- Kasha KJ, Maluszynski M (2003) Production of doubled haploids in crop plants. An introduction. In: Maluszynski M, Kasha KJ, Forster BP, Szaejko I (eds) Doubled haploid production in crop plants. A manual. Kluwer/FAO-IAEA, Dordrecht/Vienna, pp 1–4
- Katagiri K (1989) Callus induction in mulberry pollen. J Seri Sci Jpn 60:514-516
- Katagiri K, Modala V (1991) Effects of sugars and sugar alcohols on the division of mulberry pollen in tissue culture. J Seri Sci Jpn 60:514–6
- Katoh N, Iwai S (1993) Induction of haploid plants from unpollinated ovules in Nicotiana rustica. Plant Tissue Cult Lett 10(2):123–129
- Katoh N, Hagimori M, Iwai S (1993) Production of haploid plants of melon by pseudofertilized ovule culture. Plant Tissue Cult Lett 10:60–66
- Katsuo K (1969) Anther culture in tea plant (a preliminary report). Study of tea. 4:31
- Keller J (1990) Culture of unpollinated ovules, ovaries, and flower buds in some species of genus Allium and haploid induction via gynogenesis in onion (*Allium cepa* L.). Euphytica 47:241– 247
- Kiss J, Kondrak M, Törjek O, Kiss E, Gyulai G, M'azik-Tökei K et al (2001) Morphological and RAPD analysis of poplar trees of anther culture origin. Euphytica 118:213–21
- Kim JH, Moon HK, Park JI (1986) Haploid plantlet induction through anther culture of Populus maximowiczii. Res Rep Inst For Gen Korea. 22:116–21 (Korean with English summary)
- Kobayashi RS, Sinden SL Bouwkamp JC (1993) Ovule cultures of sweet potato (Ipomoea batatas) and closely related species. Plant Cell Tiss Org 32:77–82
- Kolova L, Janse J, Voorrips R (1994) Isolation and culture of apple microspores: some preliminary results. In: Schmidt H, Kellerhals M (eds) Progress in temperate fruit breeding. Kluwer Academic Publ, Dordrecht, the Netherlands, pp 409
- Kumar A, Altabella T, Taylor MA, Tiburcio AF (1997) Recent advances in polyamine research. Trends in Plant Sci 2(4):124–130
- Kunzel G, Korzuna L, Meistera A (2000) Cytologically integrated physical restriction fragment length polymorphism maps for the barley genome based on translocation breakpoints. Genetics 154:397–412

- Kuo CS (1982) The preliminary studies on culture of unfertilized ovary of rice in vitro. Acta Bot Sin 24:33–38
- Kurtar ES, Balkaya A (2010) Production of in vitro haploid plants from *in situ* induced haploid embryos in winter squash (Cucurbita maxima Duchesne ex Lam.) via irradiated pollen. Plant Cell Tiss Org 102:267–277
- Kyo M, Harada H (1986) Control of the developmental pathway of tobacco pollen in vitro. Planta 168:427–432
- Lacadena JR (1974) Spontaneous and induced parthenogenesis and androgenesis. In: Kasha KJ (ed) Haploids in higher plants advances and potential. University of Guelph, Guelph, Canada, pp 13–32
- Lakshmi Sita G, Ravindran S (1991) Gynogenic plants from ovary cultures of mulberry (Morus indica). In: Prakash J, Pierik KLM (eds) Horticulture new techniques and applications, Kluwer Academic Publishers, London, pp 225–229
- Lanaud C (1981) Production of Coffea canephora plantlets by somatic embryogenesis obtained by in vitro culture of ovules. Café Cacao Thé 25:231–236
- Last DI, Bretell RIS. (1990) Embryo yield in wheat anther culture is influenced by the choice of sugar in the culture medium. Plant Cell Rep 9:14–16
- Liang GH, Xu A, Tang H (1987) Direct generation of wheat haploids via anther culture. Crop Sci 27:336–339
- Lloyd G, McCown BH (1980) Commercially feasible micropropagation of mountain laurel, Kalmia latifolia, by use of shoot-tip culture. Proc Int Plant Prop 30:421–7
- Loureiro J, Rodriguez E, Dolezel J, Santos C (2007) Two new nuclear isolation buffers for plant DNA flow cytometry: A test with 37 species. Ann Bot 100:875–888
- Lux HL, Hermann, Wetzel C (1990) Production of haploid sugar beet (Beta vulgaris) by culturing unpollinated ovules. Plant Breed 104:177–183
- Magoon ML, Khanna KR (1963) Haploids. Caryologia 16:191-234
- Maheshwari SC, Rashid A, Tyagi AK (1980) Physiology of pollen haploid formation the current status. In: Sala F, Parisi B, Cella R, Ciferri O (eds) Plant cell cultures – Results and perspectives, North-Holland, Elsevier, pp 393–398
- Maheshwari SC, Rashid A, Tyagy AK (1982) Haploid from pollen grains-retrospect and prospect. Am J Bot 69:865–879
- Maluszynski M, Kasha KJ, Forster BP, Szarejko I (2003a) Doubled haploid production in crop plants: A manual. Kluwer, Dordrecht
- Maluszynski M, Kasha KJ, Szarejko I (2003b) Published double haploid protocols in plant species. In: Maluszynski M, Kasha KJ, Forster BP, Szarejko I (eds) Haploid production in crop plants: A manual. Kluwer, Dordrecht, pp 309–335
- Miller CO (1963) Moderne Methoden der Pflanzenanalyze. In: Linskens HF, Tracey MV (eds) Vol. 6, Springer-Verlag, Berlin, pp 194
- Mofidabadi A, Kiss J, Mázik-Tökei K, Gergácz J, Heszky LE (1995) Callus induction and haploid plant regeneration from anther culture of two poplar species. Silvae Genet 44:141–5
- Mondal TK, Parathi R (2003) Micrografting-A technique to shorten the hardening time of in vitro raised tea plants. Sri Lanka J Tea Sci (Communicated)
- Mondal TK et al (2004) Plant Cell Tissue Org, Netherlands, 76:195-254
- Montes S (1981) Obtención de callos a partir de cultivo in vitro de anteras en *Coffea rabica* L. Cult Trop 3:37–44
- Morrison RA, Evans DA (1987) Gametoclonal variation. Plant Breed Rev 5:359-391
- Muirhead KA, Horan PK, Poste G (1985) Flow cytometry: present and future Bio technol 3:337– 356
- Murashige T, Skoog F (1962) A revised medium for rapid growth and bioassays with tobacco tissue cultures. Physiol Plant 15:473–97
- Nair S, Gupta PK, Mascarenhas AF (1983) Haploid plants from in vitro anther culture of Annona squamosa Linn. Plant Cell Rep 2:198–200
- Nitsch C (1977) Culture of isolated microspore. In: Reinert J, Bajaj YPS (eds) Plant Cell Tissue Organ cen. Springer-Verlag, Berlin/Heidelberg/New York, pp. 268–278

Nitsch JP, Nitsch C (1969) Haploid plants from pollen grains. Science 163:85-7

- Nitzche W (1970) Herstellung haploider Pflanzen aus Festuca-Lolium Bastarden. Naturwissenschaft 57:199–200
- Oiyama II, Kobayashi S (1993) Haploid obtained from diploid×triploid crosses of Citrus. J Jpn Soc Hort Sci 62(1):89–93
- Okano N, Fuchinone Y (1970) Production of haploid plants by anther culture of tea in vitro. Jpn J Breed 20:63–4
- Ouyang TW, Hu H, Chuang CC, Tseng CC (1973) Induction of pollen plants from anther of *Triticum aestivum L. cultured in vitro*. Sci Sin 16:79–95
- Osolnik B, Bohanec B, Jelaska S (1993) Stimulation of androgenesis in white cabbage (Brassica oleracea var. capitata) anthers by low temperature and anther dissection. Plant Cell Tiss Org 32:241–246
- Palmer CE, Keller WA, Kasha KJ (eds) (2005) Haploids in crop improvement II. vol 56, Springer, Heidelberg, Germany
- Pandey KK, Przywara L, Sanders PM (1990) Induced parthenogenesis in kiwifruit (Actinidia deliciosa) through the use of lethally irradiated pollen. Euphytica 51:1–9
- Pedroso MC, Pais MS (1994) Induction of microspore embryogenesis in Camellia japonica cv. Elegans. Plant Cell Tiss Org 37:129–136
- Pelletier G, Ilami M (1972) Les facteurs de l'androgenèse in vitro chez Nicotiana tabacum. Z Pflanzenphysiol 68:97–114
- Perera PIP, Hocher V, Verdiil JL, Doulbeau S, Yakandawala DMD, Weerakoon, LK (2007) Unfertilised ovary: a novel explant for coconut (*Cocos nucifera* L.) somatic embryogenesis. Plant Cell Rep 26:21–28
- Pochard E, Dumas de Vaulx R (1971) La monoploidie chez le piment (*Capsicum annum* L.) Z Pflanzenzucht 65:23–46
- Powell W (1988) The influence of genotype and temperature pretreatment on anther culture response in barley (*Hordeum vulgare* L.). Plant Cell Tiss Org 12:291–297
- Powell W (1990) Environmental and genetic aspects of pollen embryogenesis. In: Bajaj YPS (ed) Biotechnology in agriculture and forestry, part I. Haploids in crop improvement, Springer, Berlin, 12:44–65
- Pratap A, Gupta SK, and Takahata Y (2009) Microsporogenesis and haploidy breeding. In: Gupta SK (ed) Biology and breeding of crucifers. CRC Press, Boca Raton, pp 293–307
- Qian C, Chen Z, Cen M, Lin M, Xu X (1982) Improvement of dedifferentiation medium in anther culture of Hevea brasiliensis Muell.-Arg. Annu Rep last Genet Acad Sin 1–19
- Raghavan V (1990) From microspore to embryo: faces of the angiosperm pollen grain. In: Nijkamp HJJ, van der Plas LH, van Hartrigik J (eds) Progress in plant cellular and molecular biology. I.A.P.T.C. Kluwer, Dordrecht, pp 213–221
- Raina SK, Iyer RD (1992) Multicell pollen proembryoid and callus formation in Tea. J Plant Crops 9:100–4
- Rajyalakshmi K, Chowdhry CN, Maheshwari N, Maheshwari SC (1995) Anther culture response in some Indian wheat cultivars and the role of polyamines in induction of haploids. Phytomorphology 45:139–145
- Reed S (2005). Embryo Rescue. In: Trigiano RN, Gray DJ (eds) Plant development and biotechnology. CRC Press, pp. 235–239
- Reinert J, Bajaj YPS (1977) Anther culture: Haploid production and its significance. In: Reinert J, Bajaj YPS (eds) Applied and fundamental aspects of plant cell, tissue and organ culture. Springer, Berlin, pp 251–267
- Saha SK, Bhattacharya NM (1992) Stimulating effect of elevated temperatures on callus production of meristemoids from pollen culture of tea (*Camellia sinensis* (L.) O. Kuntze). Indian J Exp Biol 30:83–6
- San Noeum LH (1976) Haploides d' *Hordeum vulgare* L. Par culture in vitro non fecondes. Ann Amelior Plantes 26:751–754
- San Noeum LH (1979) In vitro induction of gynogenesis in higher plants. Proc. Conf. Broadening Genet. Base Crops, Wageningen, 1978, Wageningen: Pudoc, pp 327–329

- San Noeum LH, Gelebart P (1986) Production of gynogenic haploids. In: Vasil IK (ed) Cell culture and somatic cell genetics of plants, Academic Press, New York, 3:305–322
- Sangwan-Norreel BS (1977) Androgenic stimulating factors in the anther and isolated pollen grain culture of Datura innoxia Mill. J exp Bot 28:843–852
- Sangwan-Norreel BS, Sangwan RS, Pare J (1986) Haplor die et embryogene'se provoque'e in vitro. Bull Society of Botany Fr 133. Actual Bot 4:7–39
- Saravanan M, Maria John KM, Raj Kumar R, Pius PK, Sasikumar R (2005) Genetic diversity of UPASI tea clones (*Camellia sinensis* (L.) O. Kuntze) on the basis of total catechins and their fractions, Phytochemistry. 66(5):561–565
- Seran TH, Hirimburegama K, Hirimburegama WK, Shanmugarajah V (1999) Callus formation in anther cultures of tea clones, *Camellia sinensis* (L.) O. Kuntze. J Natl Sci Found Sri Lanka 27:165–175
- Sethi M, Bose S, Kapur A, Rangaswamy NS (1992) Embryo differentiation in anther culture of mulberry. Ind J Expri Bio 30:1146–8
- Shalaby TA (2007) Factors affecting haploid induction through in vitro gynogenesis in summer squash (*Cucurbita pepo* L.). Sci Hortic 115:1–6
- Shapiro HM (1985) Practical flow cytometry. Alan R. Liss, Inc., New York
- Sharma KK, Bhojwani SS (1989) Histological and histochemical investigations of pollen embryos of *Brassica juncea* (L.) Czern. Biol Plant 31:276–279
- Sharp WR, Caldas LS, Crocomo OJ, Monaco LC, Carvalho A (1973) Production of Coffea arabica callus of three ploidy levels and subsequent morphogenesis. Phyton 31:67–74
- Shariatpanahi ME, Belogradova K, Hessamvaziri L, Heberle-Bors E, Touraev A (2006) Efficient embryogenesis and regeneration in freshly isolated and cultured wheat (*Triticum aestivum* L.) microspores without stress pretreatment. Plant Cell Rep 25:1294–1299
- Shimokado TT, Murata, Miyaji Y (1986) Formation of embryoids by anther culture of Tea. Jpn J Breed 36:282–3
- Shoukang L, Dongfeng JI, Jun Qin (1987) In vitro production of haploid plants from mulberry (Morus) anther culture. Sci Sinica 30:853–63
- Sibi ML, Kobbaissi A, Shekafandeh A (2001) Green haploid plants from unpollinated ovary culture in tetraploid wheat (Triticum durum Defs.). Euphytica 122:351–359
- Sopory SK, Jacobsen E, Wenzel G (1978) Production of monohaploid embryoids and plantlets in cultured anthers of Solanum tuberosun. Plant Sci Lett 12:47–54
- Sopory S, Munshi M (1996) Anther culture. In: Mohan JM et al (eds) In vitro haploid production in higher plants, Kluwer, Dordrecht, 1:145–176
- Srivastava P, Chaturvedi R (2008) In vitro androgenesis in tree species: An update and prospect for further research. Biotechnol Adv 26:482–491
- Srivastava P, Chaturvedi R (2011) Increased production of azadirachtin from an improved method of androgenic cultures of a medicinal tree Azadirachta indica A. Juss. Plant Signaling Behav 6(7) (In press)
- Srivastava P, Singh M, Mathur P, Chaturvedi R (2009) In vitro organogenesis and plant regeneration from unpollinated ovary: A novel explant of neem (Azadirachta indica A. Juss.). Biol Plant 53:360–364
- Stoehr MU, Zsuffa L (1990) Induction of haploids in Populus maximowiczii via embryogenic callus. Plant Cell Tiss Org 23:49–58
- Sunderland N, Collins GB, Dunwell JM (1974) The role of nuclear fusion in pollen embryogenesis of Datura innoxia Mill. Planta 117:227–41
- Sunderland N, Dunwell JM (1977) Anther and pollen culture. In: Street HE (ed) Plant tissue and cell culture. Oxford, Blackwell, pp 223–265
- Sunderland N, Roberts M (1977) New approach to pollen culture. Nature 270:236-238
- Szarejko I, Forster BP (2006) Doubled haploidy and induced mutation. Euphytica 158:359-370
- Tang FY, Tao YZ, Zhao TY, Wang GY (2006) In vitro production of haploid and double haploid plants from pollinated ovaries of maize (Zea mays). Plant Cell Tiss Org 84:233–237
- Thomas, TD, Bhatnagar AK, Razdan MK, Bhojwani SS (1999) A reproducible protocol for the production of gynogenic haploids of mulberry, *Morus alba* L. Euphytica 110:169–173

- Thomas WTB, Gertson B, Forster BP (2003) Doubled haploids in breeding. In: Maluszynski M, Kasha KJ, Forster BP, Szarejko I (eds) Doubled haploid production in crop plants: a Manual. Kluwer Academic, London, pp 337–350
- Thorthwaite JT et al (1985) A review of DNA flow cytometric preparatory and analytical methods. Immunocytochemistry in tumor diagnosis. Ed. J. Russo Martinus Nijhoff. Boston, pp 380–398
- Tiainen T (1992) The role of ethylene and reducing agents on anther culture response of tetraploid potato (*Solanum tuberosum* L.). Plant Cell Rep 10: 604–607
- Toojinda T, Baird E, Booth A, Broers L, Hayes P, Powell W, Thomas W, Vivar H, Young G (1998) Introgression of quantitative trait loci (QTLs) determining stripe rust resistance in barley: an example of marker-assisted development. Theor Appl Genet 96:123–131
- Touraev A, Ilham A, Vicente O, Heberle-Bors E (1996a) Stress induced microspore embryogenesis from tobacco microspores: an optimized system for molecular studies. Plant Cell Rep 15:561–565
- Touraev A, Indrianto A, Wratschko I, Vicente O, Heberle-Bors E (1996b) Efficient microspore embryogenesis in wheat (*Triticum aestivum* L.) induced by starvation at high temperatures. Sex Plant Reprod 9:209–215
- Touraev A, Pfosser M, Heberle-Bors E (2001) The microspore: a haploid multipurpose cell. Adv Bot Res 35:53–109
- Touraev A, Forster BP, Jain SM (eds) (2009) Advances in haploid production in higher plants. Springer, Berlin
- Tosca A, Arcara L, Frangi P (1999) Effect of genotype and season on gynogenesis efficiency in Gerbera. Plant Cell Tiss Org 59:77–80
- Uchimiya H, Kameya T, Takanashi N (1971) In vitro culture of unfertilized ovules in Solanum melongena and Zea mays. Jpn J Breeding 21:329–330
- Wan Y, Duncan DR, Rayburn AL, Petolino JF, Widholm JM (1991) The use of antimicrotubule herbicicides for the production of doubled haploid plants from anther derived maize callus. Theor Appl Genet 81:205–211
- Wan Y, Rocheford TR, Widholm JM (1992) RFLP analysis to identify putative chromosomal regions involved in the anther culture response and callus formation of maize. Theor Appl Genet 85:360–365
- Wakizuka T, Nakajima T (1975) Development of proembryo in cultured ovules of Petunia hybrida Vilm. Jap J Breed 25:161–167
- Wang CC, Chu ZC, Sun CS (1975) Induction of pollen plants of Populus. Acta Bot Sin 17:56-62
- Wang CC, Kuang BJ (1981) Induction of haploid plants from the female gametophyte of *Hordeum* vulgare. L Acta Bot Sin 23:329–330
- Wang Z, Taramino G, Yong D, Liu G, Tingey SV, Miao G-L, Wang G-L (2001) Rice ESTs with disease-resistance gene or defenseresponse gene-like sequences mapped to regions containing major resistance genes or QTLs. Mol Gen Genome 265:302–310
- Wedzony M, Forster BP, Zur I, Golemiec E, Szechynska-Hebda M, Dubas Gotebiowska G (2009) Progress in doubled haploid technology in higher plants. In: Touraev A, Forster BP, Jain SM (eds) Advances in haploid production in higher plants. Springer, Berlin, pp 1–34
- Wenzel G, Frei U, Jahoor A, Graner A, Foroughghi-Wehr B (1995) Haploids-an integral part of applied and basic research. In: Terzi M et al (eds) Current issues in plant molecular and cellular biology. Kluwer, Dordrecht, pp 127–135
- Wu BJ, Chen KC (1982) Cytological and embryological studies on haploid plant production from cultured unpollinated ovaries of *Nicotiana tabacum* L. Act Bot Sin 24:125–129
- Xu ZH (1990) Barley (*Hordeum vulgare* L.): Anther culture and the production of haploids.In Bajaj YPS (eds) Haploids in crop improvement biotechnology in agriculture and forestry. Springer-Verlag, Berlin, pp 125–175
- Xu L, Najeeb U, Tang GX, Gu HH, Zhang GQ, He Y, Zhou WJ (2007) Haploid and doubled haploid technology. Adv Bot Res 45:181–216
- Xue GR, Niu JZ (1984) A study on the induction of apple pollen plants. Acta Hortic Sin 11:161-164
- Yan H, Dong J, Zhou C, Yang, HY (1987) Regulation of in vitro parthenogenesis and somatic proliferation in sunflower by several factors. Acta Bot Sin 29(6):580–587

- Yang HY, Zhou C (1982) In vitro induction of haploid plants from unpollinated ovaries and ovule. Theor Appl Genet 6:97–104
- Zenketler M, Straub J (1979) Cytoembryological study on the process of fertilization and the development of haploid embryo of *Triticum aestivum*(2n=42) after crossing with *Hordeum bulbosum* (2n=14). Z Pflanzenzuecht 82:36–44
- Zhang YX, Lespinasse Y, Chevreau E (1990) Induction of haploidy in fruit trees. Acta Hort 280:293–304
- Zhang YX, Lespinasse Y (1991) Pollination with gamma irradiation pollen and development of fruits, seeds and parthenogenetic plants in apple. Euphytica 54:101–109
- Zhou C, Yang HY (1981) Studies on the in vitro induction of callus from embryo sacs of rice. Hereditas (Beijing) 3(5):10–12
- Zhou H (1996) Genetics of green plant regeneration from anther culture in cereals. In: Jain SM, Sopory SK, Veilleux RE (eds) Vitro haploid production in higher plants vol. 2. Kluwer Academic Publishers, The Netherlands, pp 169–187
- Zou CJ, Li PF (1981) Induction of pollen plants of grape (Vitis vinifera L.). Acta Bot Sin 1:79-81
- Zhu ZC, Wu HS (1979) In vitro induction of haploid plantlets from the unpollinated ovaries of *Triticum aestivum* and Nicotiana tabacum. Acta Genet Sin 6:181–183
- Zhu XY, Wang RL, Liang Y (1980) Induction of poplar pollen plants. Sci Silvae Sin 16:190-7
- Zhu ZC, Wu HS, An QK, Liu ZY (1981) Induction of haploid plantlets from unpollinated ovaries of *Triticum aestivum* cultured in vitro. Acta Genet Sin 8:386–390

Index

A

Abiotic stress effects of, 141, 154, 157, 161, 165, 213, 367 programmed cell death (PCD), 124, 143, 144 role of, 138, 146 Abiotic stress tolerance biochemical, 77, 78, 114, 116, 118, 121, 124, 125, 127, 130, 132, 143, 146, 301, 311, 357, 374 antioxidants, 143, 146, 301, 374 cryoprotectant, 127 host's metabolism (metabolites), 114, 116, 118, 121, 124, 125, 127, 130, 132, 143, 357 rhizosphere, 77, 78, 311 morphological, 121-123, 292, 366 hinder transpiration, 122, 292, 366 sunken stomata, 121, 123, 366 physiological, 113, 116, 118, 121, 123, 124, 127, 130, 142-145, 158, 299, 323, 350, 356, 366, 368 growth control, 299 hormone regulation, 121, 144 host defense mechanism, 116, 118, 368 PCD, 124, 143-145 proteins, 43, 113, 121, 130, 142, 158, 323, 350, 356, see also Metallothioneins (MTs) transcription factors, 123, 127, 130 water homeostasis, 366 Abiotic stressors chemical factors, 2, 14, 19, 20, 48, 105, 108, 138, 142, 154, 156, 157, 160, 213, 286, 310, 342, 347, 349, 361 heavy metals, 108, 154, 156, 157, 160, 286, 347, 349, 361 pesticides and herbicides, 286, 361

salinity, 2, 19, 20, 138, 142, 156, 213, 286 soil and water pH, 48, 105, 310, 342 toxins, 14 environmental/ecological factors, 38, 47, 50, 51, 157, 362 environmental pollutants, 47, 50, 51, 157, 362 natural calamities, 38, 47, 50, 51 physical factors, 2, 19, 20, 38, 45, 47, 127, 138, 154, 156, 163, 164, 213, 218, 286, 287, 300 drought, 2, 156, 213, 218, 287, 300 flood, 218 ionizing radiation (gamma rays), 154, 163, 164 non ionizing radiation (UV/IR), 138, 154, 164, 286 temperature (low/extreme), 2, 19, 20, 45, 127, 138, 218, 300 wind, 38, 47 Abscisic acid (ABA) accumulation of, 43, 123 as growth regulators, 184 biosynthesis of, 43, 123 defoliation, 123 drought stress, 124 Agricultural biotechnology, 6, 213, 280 Agrobacterium tumefaciens, 219 Agrochemicals, 1, 187, 231, 232, 240, 241, 247 Agronomy, 146 γ-Aminobutyric acid (GABA), 118 D-Aminolevulinic acid (ALA), 320 Andaman and Nicobar, 36, 50, 57 Androgenesis, 175, 178, 182, 183, 185, 186, 188, 193, 214, 219, 221, see also

Androgenic haploids

Androgenic haploids anther culture, 173-178 chemical treatment, 174 conditions for, 177 direct embryogenesis, 175 doubled haploids, see Doubled haploids explants, 176 mutagenesis/genetic engineering in, 178 pollen/microspore culture, 173, 178, 192, 196 embryos and callus, 178 factors affecting, 178 microspore division, 192, 196 Antioxidants as free radical scavenger, 44, 296, 297, 299, 324, 348, 363, 367, 368, 373, 374 as stress marker, see Glutathione (GSH) in defense mechanism, 44, 127, 144, 154, 157, 162, 220, 324, 363, 367, 368, 373 sugar as, 290, 295, 301, see also Carbohydrates tripeptidthiol, 127 vitamin C, 127, 193 vitamin E, 127 Arsenate stress, see Arsenic toxicity Arsenic (As) detoxification of, in plants, 352 in water pollution, 347 toxicity of, see Arsenic toxicity uptake and metabolism, 350, 351 uptake by plants, 347, 349 arsenate (AsV), 347, 349 arsenite (AsIII), 347, 349 methylated arsenic, 347, 349 Arsenic tolerance in humans, 354 in plants, 354-357 classical/molecular breeding, 354, 355 genetic engineering of, 354, 355 genetic mapping of, 355 mechanism of, 355-357 Arsenic toxicity in plants, 348-350, 354 hyperaccumulator, 348 phytoremediation, 354 ROS, 348 stress resistances, 354 uptake and transport, 349, 350 Ascorbate peroxidase (APX) H₂O₂ scavenger, 324, 368

in chloroplast, 368 maintains plant genome integrity, 155 splicing, 368 Ascorbate–glutathione (AsA–GSH) cycle, 363, 370, 373, 374

B

Bacillus subtilis, 5, 293 6-Benzylaminopurine (BAP), 179 Biocontrol agents biopesticide, 188, see also Nematode trapping fungi Biofuels, 1 Biotic stress, 141, 142, 146, 271, 278, 280 Biotic stress tolerances, 137, 141, 144, 145, 271, 279 Biotic stressors diseases, 138 pathogens, 233-235, 271, 277, 278, 280 bacteria, 235 fungi, 233, 234 parasites, 271, 277, 278, 280 Boron (B) as trace element, 333 boric acids, 334, 338 essentiality of, in plants, 333, 335 in soils, 334 Boron tolerance efflux transporters in, 341 aquaporins, 341 genetics of (Bor genes), 340 AtBor1, 340 AtBor4, 340 HvBor2, 340 OsBor2, 340 TaBor2, 340 in leaves, 341 mechanism of, 342 physiology of, 337, 339 Boron toxicity causes of, 335 salinity, 342 symptoms of, 335, 336 Breeders, 6, 10, 171, 200, 213, 219

С

Cadmium (Cd) as environmental pollutant, 361–363 in microbes, 364 in plants, 350, 362–365 accumulation of, 363–365 concentration of, 362 uptake and transport, 350, 363–365

Index

ROS. see Oxidative stress and SOD activity solubility, 362 Cadmium tolerance mechanism of 367-370 antioxidants, see Antioxidants CAT activity, 368, 369 glutathione reductase (GR) activity, 370 minimized ROS production, 367, 368 Cadmium toxicity detoxification mechanism, see Cadmium tolerance in humans and animals, 362, 363 oxidative stress, 366, 367 photosynthesis inhibition, 362, 363, 365, 366 symptoms of, 362, 364 Carbohydrates as antioxidants, 297 as biomolecule for cellular metabolism, 255, 285, 300 as osmoprotectants, 290, 291, 296 as signals, 294, 295 in plant biomass, 300 stress-related sugars, metabolism of, 294, 299 sugar vitrification, 290 Carbohydrates, stress related disaccharides, 287 fructans, 288-290, 292, 293 raffinose, 288, 292 Carbon assimilation, 300, 317, 318 Carbon nanomaterials, see Nanotechnology Carbon sequestration, 49, 50 Catalase (CAT) as H₂O₂ scavenger, 369 in Cd stress, see Cadmium tolerance isozymes of, 369 Cation exchange capacity (CEC), 322 Cereals, 157, 173, 174, 178, 183, 214, 215, 217, 221, 223, 314 Ceylon tea, 94 Chemical nematicides, 272 Chilling stress, see Abiotic stressors Chromium (Cr) as water pollutant, 310 Cr(III), 309, 310 Cr(VI), 309, 310, 315 environmental concentration of, in India, 310 immobilization of, 311, 312 interacting with plant metabolism, 310, 311, 318

mechanism of, 310, 311 photoinhibition, 318 occurrences, 309 man-made, 309 natural. 309 oxidative damage in plants, 315, 321, see also Lipid peroxidation redox state, 310 ROS production in plants, see Oxidative stress tolerance mechanism in plants, 316, 323, 324 Cr detoxification, 316, 323, see also Metallothioneins (MTs) enhanced SOD activity, 324 toxicity of, 310, see also Chromium toxicity translocation in plants, 313 transportation index of, 313 uptake and accumulation in plants, 310-313 Chromium reactivity, 322 Chromium toxicity mechanism of, 321 organic acids in, 322 phytotoxicity effect, 314, 315, 317, 319-321 decreased chlorophyll synthesis, 319, 320 impaired carbon assimilation, 317 impaired nitrogen assimilation, 320 intrusion in chlorophyll biosynthesis, 320 redox imbalance, 321 symptoms of, 314-316 Circadian proteins, 164 Climate change, 2, 3, 30, 45, 47, 58, 153 Cold stress, see Abiotic stressors Crop genetics, 3, 5-7, 19, 22 Crop improvement nanomaterials for, see Nanotechnology Crop plants, 124, 132, 142, 174, 177, 218, 221, 259, 343, 347, 348, 363, 364, 375 Crop productivity, 95, 153, 156, 157, 160, 165, 361 Cryptovivipary, 40 Cytoplasmic male sterility (CMS), 287

D

DAYCENT model, 72 2, 4-Dichlorophenoxyacetic acid (2, 4 D), 179, 184, 188, 190 Diploidization, 199–201, *see also* Doubled haploids Disaccharides, *see* Carbohydrates DNA repair, 154–156, 158, 160–163, 165 DNDC model, 72 Doubled haploids, 172–175, 195, 199, 200, 202, 203 Drought, 2, 13, 300, 343 Drought tolerance, 4, 14, 18, 26, 30, 121, 122, 157, 238, 292

Е

Ecosystems, 1, 47, 72, 95, 157, 289 Efflux transporter, *see* Boron tolerance Embryo rescue hybrids production, 180 Embryogenesis callus induction, 178, 188, 190 Environmental pollution, air, 156, 157 by heavy metals, 156, 157, 361

F

Flow cytometry, 192, 197, 198, 217 Fructans, *see* Carbohydrates Fungal toxins, 272

G

Gene expression, 140, 144, 145, 153, 155, 163, 220, 236, 238, 247, 265, 294, 323, 326, 336 Genetic engineering, 77, 132, 153, 178, 213, 215, 219, 220, 241, 279, 354, 355, 375 Genetic transformation, see also Genetic engineering Agrobacterium-mediated, 123, 215, 216, 219, 237, see also Agrobacterium tumefaciens androgenesis, 214, 218, 220 biolistics gun, 237 nanotechnology, see Nanotechnology protoplast transformation/electroportion, 219 transgenic plants, 214, 218, 220 using plasmids, 216, 238 Genotoxic adaptation, 155, 156 Genotoxic stress, see also Genotoxic adaptation adaptive response circadian proteins, 164 gene expression, 154, 156, 163-165 signaling pathways, 155, 156, 158, 164

aluminum pollution, 156 cancer in human, 154 clastogenic adaptation, 155 effects of, 164 in Arabidopsis mutants, 164 genotoxic agents, 154, 156, 157, 163, 164 GST activity, 370 impact of, 165 in DNA repair mechanism, 155, 158, 159, 164 base excision repair (BER), 155, 159, 164 nucleotide excision repair (NER), 158 oxidative injury, 158 perception of, 154 programmed cell death, 154 salvcilic acid (SA) pretreatment, 156 Genotype-environment interaction, 131 Germplasm, 4 GIS technology, 92 Global warming, 46, 47, 49, 299 Global warming potential (GWP) carbon dioxide, 67, 68, 71 methane, 67, 70, 71, 73-75, 82 nitrous oxide, 67, 70, 71 Glutathione (GSH), 286, 363, 370, 372 Gradient centrifugation, 178 Green revolution, 3, 4, 220 Greenhouse gas (GHG), 65-67, 73, 74 Growth additives, 184 Growth regulators ABA, see Abscisic acid (ABA) auxins (IAA), 137, 142, 176, 179, 184, 262 cytokinins, 176, 179, 184 2, 4 D, see 2, 4-Dichlorophenoxyacetic acid giberrellins, 137, 184 role of, 184 Gynogenesis

H

cold treatment, 186

H₂O₂ scavengers ascorbate peroxidase, *see* Ascorbate peroxidase (APX) catalase, *see* Catalase (CAT) glutathione (GSH), 143, 286 glutathione-S-transferases (GST), 370, 373 Haploids, 171, 180, 203 Haploids productions application of, 201–203, 217 genetic stability, 201, 203 in crop improvement, 201, 203

in gametoclonal variation, 201 in hybrid production, 201 in plant breeding, homozygous lines, 203, 217 in transgenic plants, 202 mutation induction, 202 diploidization of haploids, 174, 175, 199, 200 factors affecting, 181 genetic transformation, 172 in plant breeding, 200 in vitro androgenesis techniques, 172-174, see also Androgenic haploids in vitro gynogenesis, 172, 173, 178, 179 mutagenesis, 172 ovule androgenesis, 172 parthenogenesis, 172, 173, 178 wide hybridization, 174, 180, see also Wide hybridization Hardening of root and shoot system, 200 Heavy metals arsenic (As), see Arsenic cadmium (Cd), 157, 162, 261, 320, 350, 361, 362 chromium (Cr), see Chromium copper (Cu), 108, 157, 362 genotoxic effects of, 157 lead (Pb), 108, 157, 261, 320 mercury, 261, 320, 362 nickel (Ni), 108, 362 High throughput screening DNA microarray, 121, 130, 165, 355 protein profiling, 165 Homozygosity, see Haploids productions Homozygous diploids, see Doubled haploids

I

Indole-3-acetic acid (IAA), *see* Growth regulators Intellectual property (IP) rights, 6, 7, 9–11 International Rice Research Institute (IRRI), 77

L

Lipid peroxidation, *see also* Oxidative stress heavy metals, 322, 348, 363, 367 arsenic, 348 cadmium, 363, 367 chromium, 322 indicator of free radicals, 249, 299, 321, 363, 367 malondialdehyde (MDA), 297, 321

М

Magnetic nanoparticles metabolic changes in plants, 237 scope of, 237 treatment of plant diseases, 237 Manganese toxicity symptoms of, 104 Mangroves adaptations anatomical, 40 morphological, 39 physiological, 40 as carbon factories, 49 ecosystem, 35, 42 occurrences, 35, 36 organic osmolytes of, 43 reproduction methods in. 40 root systems in, pneumatophores, 35, 39, 40, 48 salt controlling/regulatory mechanism, 40, 42 - 45salt accumulation, 42 salt exclusion. 42 salt excretion, 42, 44 salt secretion, 44 salt tolerance, 40, 43-45 Marker assisted selection (MAS), 203 Metallothioneins (MTs) as chelating agent, 323, 324 Cr detoxification in plants, 310, 323 DNA repair machinery, 158 heavy metal detoxification, 158 MT mRNA transcription, 323 MT3 gene expression, 158, 323 ROS scavengers, 157, 158 zinc transport in animal cell, 158 Methane (CH₄) emission, 66, 67, 69, 78, 81 Methane transport capacity (MTC), 80, 81, 83 Methyl methane sulfonate (MMS), 164 Michaelis-Menten kinetics, 310, 350 Mineral nutrition, 315, 316 Mitigation of GHGs agriculture ecosystems, 72 Mycoparasitism, 278

N

Nanobiopesticide, *see* Nanotechnology Nanobiotechnology, 246–248, 250, 264, 265 Nanomaterials/nanoparticles crop improvement, 234, 235, 257–262, 264 carbon, 235, 257 gold, 260, 261

magnetic, 258, 259 nanosilica, 234 palladium, 261 silver, 234, 262, see also Silver nanoparticles titanium dioxide (TiO₂), 235, 262, 264 zinc oxide (ZnO), 262 engineered, 248-250 carbon-based, 249 composites, 250 dendrimers, 249 metal-based, 249 incidental. 248 natural, 248 plants interaction, 255-257, 264 genotoxicity of, 264 nanoparticles delivery, 255-257 nanoparticles synthesis, 257 phytotoxicity of, 264 Nanotechnology in agriculture, 231-233, 238, 240, 247 agrochemicals (nanoformulated), 231, 232 crop protection, 231, 232 detection of plants disease, 232, 240 enhance nutrients assimilation, 232 nanodevices/nanosensors, 231, 247, see also Nanomaterials/nanoparticles phytopathogens control, 233 transgenic plants, 238 in medicine/biomedical, 232-235, 238, 241 antibacterial agent, 233-235 antifungal agent, 233-235 drug delivery system, 232, 233, 238 viral diagnostics, 241 nano/microfabrication, 247 Nanotubes/nanotransporters, see Nanomaterials/nanoparticles Nematode trapping fungi adhesive traps and spores, 273 applications of, 276 Royal 300, 276 Royal 350, 276 as biocontrol agents, 272-275, 279 classification of (phylogenetic studies), 274 crop management, 275 fungistasis, 275 hyphomycetes, 273 improving biocontrol potential of, 280 genetic engineering, 280 virulence factors, 280 morphology of, 273 phase shifting, 273

root endophytes, 279 stages of nematodes, 272 Nematode-fungal interaction, see Nematode trapping fungi Nitrification inhibitors S-benzylisothiouronium butanoate (SBTbutanoate), 70 S-benzylisothiouronium furoate (SBTfuroate), 70 dicyandiamide (DCD), 70 3,4-dimethylpyrazole phosphate (DMPP), 70 nitrapyrin, 70 time of application, 70 Nitrous oxide (N₂O) emission, 68-70 Northern blot, 217

0

Organic osmolytes, 43 Osmoprotectants, see Carbohydrates Oxidative stress biomarker in, see Peroxidises cellular redox homeostasis, 142, 287 GR and GSH activity, 324, 370, 373 Haber-Weiss-type reaction, 371 lipid peroxidation, 145, 322 metal-induced, 157, 322, 325, 326, 348, 362, 366, 367, 374 arsenate, 348 cadmium, 362, 366, 367, 374 chromium, 157, 322, 325, 326 lead, 157 ROS generation, 44, 157 sugar starvation, 287, 294 telomerase role in, 162

P

Peroxidases, 157, 264, 286, 298, 299, 325, 326, 367, 371, 373 Pest control, 2, 12-14, 18 Photosystems chlorophyll, 318, 367 heavy metals, 320, 322, 366 oxidative stress, 362, 366 PSI, 366 **PSII**, 366 Phytochelatin synthase, 324 Phytochelatins (PCs), 310, 323, 324 Phytonematodes as biotic stress factors, 271 chlorosis, 272 cyst nematodes, 272, 280 life cycle, 280

Index

motility, 272 RNA interference, 280 root knots, 271, 272, see also Nematode trapping fungi sensory system, 272 Plant biotechnology, 24, 219, 240 Plant breeding, 165, 203, 355 Plant development, 96, 126, 156, 183, 292, 300 Plant disease control, see Nanotechnology Plant-pathogen interaction, 264 Ploidy levels, 190, 197, 199, 217 Polyamines (PAs) in plants, 114, 116-118 as growth regulators, 114 biosynthesis of, 114 catabolism of, 114, 117, 118 concentration, 114 electrostatic-covalent interactions, 116 enzymes activity, 114 PAs reservoirs, 116 in stress tolerances, 114, 118, 120, 123, 125-127, 129, 130 cold stress, 126, 127, 129, 130 drought stress, 118, 123 pathogen attack, 114 salt stress, 120, 125 Polyethylene glycol (PEG) mediated gene transfer, 237 osmotic stress, 120, 122, 123, 159, 160, 162 Polyvinyl pyrrolidone (PVP), 195, 198 Polyvinylidene fluoride (PVDF), 199 Poverty, 1, 17 Productivity and yield, 98, 103, see also Crop productivity

R

Recombinant DNA, 6 Regeneration androgenesis, 173, 174, 177, 181, 189, 190, 192, 197 androgenic embryogenesis, 183, 194, 197, 200, 214 genetic engineering, 217–219 gynogenic embryogenesis, 183 rate of, 47 Remote sensing, 50, 51, 55, 57, 91–93, 95, 96, 98 Rice, 69, 74, 78, 79, 91, 293, 301, 347, 348, 364 Roots endophytes, *see* Nematode trapping fungi heavy metals toxicity in, 316, 322, 364 properties, 79 ultra-filtration in, 42, *see also* salt exclusion ROS homeostasis, 286, 287 mediated toxicity, *see* Oxidative stress scavenging systems, 120, 121, 324, 368–372 enzymatic, 121, 324, 368–371 non-enzymatic, 120, 372, *see also* AsA–GSH cycle signaling, 138, 142, 161, 294 stress, *see* Oxidative stress

S

Salicylic acid (SA) biosynthesis of, 139 discovery of, 137 endogenous applications of, 141, 143 exogenous applications of, 142, 144, 146 growth regulator in plants, 137, 142 H₂O₂ upregulation, 139 in plant defence mechanism, 141, 142, 144, 145 redox homeostasis, 142 ROS signaling, 142 signaling in gene expression and regulation, 145 volatile MeSA, plant signaling, 140 Salinity, 342 Salt stress in seeds, 161 mangroves, 41 Salt tolerance, see Abiotic stress tolerance SDS-PAGE, 218 Silver nanoparticles antibacterial agent, 233, 261 antifungal agent, 233, 261 optical studies, 252 Small regulatory RNAs, 154, 163 Southern blot, 217, 218 Sri Lanka climatic condition of, 91, 93 geographical location of, 91, 93 rice plantations, 91 tea plantations, 91, 93, 94, 97 Ceylon tea, 94 GIS and remote sensing, see Remote sensing and GIS technology in Ratnapura, 93 pruning cycle, 97 Stress resistance, 238, 294, 295, 348, see also Abiotic and biotic stressors

Stress severity, 131
Stress tolerance, *see* Abiotic and biotic stress tolerance
Stress tolerant crops, field trials of, 26, 299, 301
Superoxide dismutase (SOD) activity in plants Cd stress tolerance, 367
detoxifying process of, 367, 368, 371
first line of defence, 325, 367, 371
Sod1 gene, 44
superoxide radicals (O₂⁻) scavenger, 44, 286, 295, 324, 348, 371

Т

Technology adoption, 12 Telomerase in DNA repair mechanism, 162, 163 transcription activity (TERT), 162, 163 prevent aging, in animal cell, 162 restoration of, 163 seed aging, 163 Temperature, *see* Abiotic stressors Tolerance, *see* Stress tolerance Totipotency, 219 Toxicity, *see* Heavy metals Transmission electron microscopy (TEM), 251 U UV spectroscopy, 251, 252

V

Vegetation indices, 95, 96 Vivipary, 38, 39, 41, *see also* Mangroves Vivipary, advantages of, 41

W

Western blot, 218 Wide hybridization early embryogenesis, 172 embryo rescue, 180 in haploid production, 174, 180 in plant breeding programmes, 180 under stress condition, 220 Woody tree, 174, 184, 186, 195, 291

Х

Xenobiotics detoxification of, 372, *see also* Glutathione (GSH)

Z

Zn deficiency chlorosis, 103